United States Securities and Exchange Commission

Washington, D.C. 20549

Form 10-Q

QUARTERLY REPORT PURSUANT TO SECTION 13 OR 15 (d) OF THE SECURITIES AND EXCHANGE ACT OF 1934

For the quarterly period ended June 30, 2005

OR

☐ TRANSITION REPORT PURSUANT TO SECTION 13 OR 15 (d) OF THE SECURITIES AND EXCHANGE ACT OF 1934

Commission File Number 0-27517

GAIAM, INC.

(Exact name of registrant as specified in its charter)

COLORADO

(State or other jurisdiction of incorporation or organization)

84-1113527 (I.R.S. Employer

(I.R.S. Employer Identification No.)

360 INTERLOCKEN BLVD., BROOMFIELD, COLORADO 80021

(Address of principal executive offices)

(303) 222-3600

(Registrant's telephone number, including area code)

Indicate by check mark whether the registrant: (1) has filed all reports required to be filed by Section 13 or 15 (d) of the Securities and Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports) and (2) has been subject to such filing requirements for the past 90 days.

YES

■ NO □

Indicate by check mark whether the registrant is an accelerated filer (as defined in Rule 12b-2 of the Exchange Act).

YES 🗷 NO 🗆

Indicate the number of shares outstanding of each of the issuer's classes of common stock, as of the latest practicable date:

Shares outstanding as of August 8, 2005
12,301,435
5,400,000

INDEX TO FORM 10-O

PART I. FINANCIAL INFORMATION

Condensed Consolidated Financial Statements (Unaudited)

Condensed Consolidated Balance Sheets at June 30, 2005 and December 31, 2004

Condensed Consolidated Statements of Operations for the three months ended June 30, 2005 and 2004

Condensed Consolidated Statements of Operations for the six months ended June 30, 2005 and 2004

Condensed Consolidated Statements of Cash Flows for the six months ended June 30, 2005 and 2004

Notes to Interim Condensed Consolidated Financial Statements

<u>Item 3.</u> <u>Quantitative and Qualitative Disclosures About Market Risk</u>

Item 4. Controls and Procedures

PART II. OTHER INFORMATION

Item 1. Legal Proceedings

<u>Item 2.</u> <u>Unregistered Sales of Equity Securities and Use of Proceeds</u>

<u>Item 3.</u> <u>Defaults Upon Senior Securities</u>

Item 4. Submission of Matters to a Vote of Security Holders

Item 5. Other Information

<u>Item 6.</u> <u>Exhibits</u>

SIGNATURES

FORWARD-LOOKING STATEMENTS

This report may contain forward-looking statements that involve risks and uncertainties. When used in this discussion, the words "anticipate," "plan," "estimate," "expect," "strive," "future," "intend" and similar expressions as they relate to Gaiam or its management are intended to identify such forward-looking statements. Gaiam's actual results could differ materially from the results anticipated in these forward-looking statements as a result of certain factors set forth under "Management's Discussion and Analysis of Financial Condition and Results of Operations," "Market Risk" and elsewhere in this report. Risks and uncertainties that could cause actual results to differ include, without limitation, general economic conditions, competition, loss of key personnel, pricing, brand reputation, consumer trends, acquisitions, new initiatives undertaken by Gaiam, security and information systems, legal liability for website content, merchandise supply problems, failure of third parties to provide adequate service, our reliance on centralized customer service, overstocks and merchandise returns, our reliance on a centralized fulfillment center, increases in postage and shipping costs, E-commerce trends, future Internet related taxes, control of Gaiam by its founder, fluctuations in quarterly operating results, customer interest in our products, the effect of government regulation and other risks and uncertainties included in Gaiam's filings with the Securities and Exchange Commission. We caution you that no forward-looking statement is a guarantee of future performance, and you should not place undue reliance on these forward-looking statements which reflect our management's view only as of the date of this report. We undertake no obligation to update any forward-looking information.

2

GAIAM, INC. CONDENSED CONSOLIDATED BALANCE SHEETS

(In thousands, except share information)

Assets Current assets: Cash and cash equivalents \$ Accounts receivable, net Income tax and other receivables Inventory, less allowances Deferred advertising costs Deferred tax assets Other current assets Total current assets	(Unaudited)	2004
Cash and cash equivalents \$ Accounts receivable, net Income tax and other receivables Inventory, less allowances Deferred advertising costs Deferred tax assets Other current assets	(
Accounts receivable, net Income tax and other receivables Inventory, less allowances Deferred advertising costs Deferred tax assets Other current assets		
Income tax and other receivables Inventory, less allowances Deferred advertising costs Deferred tax assets Other current assets	12,259	\$ 10,439
Inventory, less allowances Deferred advertising costs Deferred tax assets Other current assets	8,129	13,914
Deferred advertising costs Deferred tax assets Other current assets	3,023	3,000
Deferred tax assets Other current assets	15,594	16,503
Other current assets	1,994	2,635
	715	1,145
otal current assets	1,462	 1,324
	43,176	 48,960
Property and equipment, net	7,128	7,857
nvestments	7,865	7,865
Capitalized production costs, net	5,326	5,457
Media library, net	5,060	5,427
Goodwill and other intangibles	9,537	9,757
Non-current deferred tax assets	3,685	2,657
Other assets	431	307
Total assets \$	82,208	\$ 88,287
iabilities and stockholders' equity		
Current liabilities:		
Accounts payable \$		

Accomed lists tries ble	3,980	3,688
Total current liabilities	12,418	17,472
Minority interest	4,373	4,469
Commitments and contingencies		
Stockholders' equity:		
Class A common stock, \$.0001 par value, 150,000,000 shares authorized, 9,420,118 and		
9,411,897 shares issued and outstanding at June 30, 2005 and December 31, 2004,		
respectively	1	1
Class B common stock, \$.0001 par value, 50,000,000 shares authorized, 5,400,000 issued and		
outstanding at June 30, 2005 and December 31, 2004	1	1
Additional paid-in capital	54,982	54,933
Accumulated other comprehensive income	522	850
Retained earnings	9,911	 10,561
Total stockholders' equity	65,417	66,346
Total liabilities and stockholders' equity	\$ 82,208	\$ 88,287

 $See\ accompanying\ notes\ to\ the\ condensed\ consolidated\ financial\ statements.$

3

GAIAM, INC. CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited)

(In thousands, except per share amounts)

	For the Three Months En June 30,			
		2005		2004
Net revenue	\$	21,706	\$	17,031
Cost of goods sold		11,149		8,491
Gross profit		10,557		8,540
Expenses:				
Selling and operating		9,935		10,171
Corporate, general and administration		1,893		2,003
Total expenses		11,828		12,174
Loss from operations		(1,271)		(3,634)
Other income (expense)		(53)		28
Interest income		45		48
Total other income (expense)		(8)		76
Loss before income taxes and minority interest		(1,279)		(3,558)
Income tax benefit		(532)		(1,297)
Minority interest in net (income) loss of consolidated subsidiaries, net of tax		(19)		46
Net loss	\$	(766)	\$	(2,215)
Net loss per share:				
Basic	\$	(0.05)	\$	(0.15)
Diluted	\$	(0.05)	\$	(0.15)
Shares used in computing net loss per share:				
Basic		14,820		14,686
Diluted		14,820		14,686

 $See\ accompanying\ notes\ to\ the\ condensed\ consolidated\ financial\ statements.$

	For the Six Months Ende June 30,			
	 2005			
Net revenue	\$ 48,030	\$ 40,806		
Cost of goods sold	23,724	19,590		
Gross profit	24,306	21,216		
Expenses:				
Selling and operating	21,625	21,128		
Corporate, general and administration	3,665	4,051		
Total expenses	25,290	25,179		
Loss from operations	(984)	(3,963)		
Other income	38	52		
Interest income	 70	80		
Total other income	 108	132		
Loss before income taxes and minority interest	(876)	(3,831)		
Income tax benefit	(402)	(1,413)		
Minority interest in net income of consolidated subsidiaries, net of tax	(176)	(126)		
Net loss	\$ (650)	\$ (2,544)		
Net loss per share:				
Basic	\$ (0.04)	\$ (0.17)		
Diluted	\$ (0.04)			
Shares used in computing net loss per share:				
Basic	14,820	14,650		
Diluted	14,820	14,650		

See accompanying notes to the condensed consolidated financial statements.

5

GAIAM, INC. CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited – In thousands)

For the Six Months Ended June 30, 2005 2004 Operating activities Net loss \$ (650) \$ (2,544)Adjustments to reconcile net loss to net cash provided by operating activities: Depreciation 1,145 1,472 Amortization 802 835 Stock compensation 36 Minority interest in consolidated subsidiaries 176 126 Deferred income tax expense 382 221 Changes in operating assets and liabilities, net of effects from acquisitions: 5,620 9,279 Accounts receivable Inventory 663 (12)Deferred advertising costs 641 (100)Other current assets (150)(91) 205 Other assets (145)Accounts payable (3,882)(4,589)Accrued liabilities (504)(602)Income taxes payable (1,312)180 Net cash provided by operating activities 2,786 4,416 **Investing activities** Purchase of property, equipment and media rights (651)(767)Net cash used in investing activities (767)(651)Financing activities Principal payments on capital leases (37)

Proceeds from issuance of common stock Net cash provided by financing activities		482 385
Effects of exchange rates on cash and cash equivalents	(199)	37
Net change in cash and cash equivalents	1,820	4,167
Cash and cash equivalents at beginning of period	10,439	8,384
Cash and cash equivalents at end of period	\$ 12,259	\$ 12,551
Supplemental cash flow information		
Interest paid	\$ _	\$ —
Income taxes paid	560	_

See accompanying notes to the condensed consolidated financial statements

6

Gaiam, Inc.

Notes to Interim Condensed Consolidated Financial Statements

(Unaudited)

June 30, 2005

1. Interim Condensed Consolidated Financial Statements

Organization and Nature of Operations

Gaiam, Inc. is a lifestyle media company providing a broad selection of content, information, and products to individuals who value personal development, natural health, ecological lifestyles and responsible, inspirational media. Gaiam was incorporated under the laws of the State of Colorado on July 7, 1988.

The accompanying condensed consolidated financial statements include the accounts of Gaiam and its subsidiaries in which Gaiam's ownership is greater than 50% and the subsidiary is considered to be under Gaiam's control. All material intercompany accounts and transaction balances have been eliminated in consolidation.

Preparation of Interim Condensed Consolidated Financial Statements

The interim condensed consolidated financial statements included herein have been prepared by the management of Gaiam pursuant to the rules and regulations of the United States Securities and Exchange Commission, and, in the opinion of management, contain all adjustments (consisting of only normal recurring adjustments) necessary to present fairly Gaiam's consolidated financial position as of June 30, 2005, the interim results of operations for the three and six months ended June 30, 2005 and 2004, and cash flows for the six months ended June 30, 2005 and 2004. These interim statements have not been audited. The balance sheet as of December 31, 2004 was derived from Gaiam's audited consolidated financial statements included in Gaiam's annual report on Form 10-K.

Certain information and footnote disclosures normally included in consolidated financial statements prepared in accordance with accounting principles generally accepted in the United States have been condensed or omitted pursuant to such rules and regulations. Accounting policies followed by Gaiam are described in Note 1 to the audited financial statements for the fiscal year ended December 31, 2004 included in Gaiam's annual report on Form 10-K. The consolidated financial statements contained herein should be read in conjunction with Gaiam's audited financial statements, including the notes thereto, for the year ended December 31, 2004.

The consolidated financial position, results of operations and cash flows for the interim periods disclosed in this report are not necessarily indicative of future financial results.

Use of Estimates

The preparation of the consolidated financial statements in conformity with accounting principles generally accepted in the United States requires management to make estimates and assumptions that affect the reported amounts of assets, liabilities, revenue and expenses and disclosure of contingent assets and liabilities at the date of the consolidated financial statements. Actual results could differ from those estimates.

Reclassifications

Certain reclassifications have been made to prior period amounts to conform to the current period presentations.

In December 2004, the FASB issued SFAS No. 123 (revised), *Share-Based Payment*, which supersedes Accounting Principles Board ("APB") No. 25, *Accounting for Stock Issued to Employees* ("APB No. 25"), and its related implementation guidance. Under SFAS No. 123 (revised), all share-based payments would be treated as other forms of compensation by recognizing the costs, generally measured as the fair value at the date of grant, in the statement of operations. Gaiam will adopt SFAS No. 123 (revised) for its fiscal year beginning January 1, 2006, in accordance with rules adopted by the Securities and Exchange Commission on April 14, 2005. Management expects that the impact of the adoption of SFAS No. 123 (revised) will be that the share-based payment expense amounts historically disclosed as required by SFAS No. 123 will now be recognized as an expense on the statement of operations.

Stock-Based Compensation

Gaiam accounts for its stock-based compensation arrangements under the provisions of APB No. 25 and related interpretations, including FASB Interpretation No. 44, *Accounting for Certain Transactions Involving Stock Compensation*, rather than the alternative fair value accounting allowed by SFAS No. 123, *Accounting for Stock Based Compensation*. Accordingly, no compensation expense is recognized in Gaiam's consolidated financial statements in connection with stock options granted to employees with exercise prices not less than fair value. Deferred compensation for options granted to employees is determined as the difference between the fair market value of Gaiam's common stock on the date options were granted and the exercise price. For purposes of this pro-forma disclosure, the estimated fair value of options is assumed to be amortized to expense over the options' vesting periods.

Compensation expense for options granted to non-employees has been determined in accordance with SFAS No. 123 as the fair value of the consideration received or the fair value of the equity instruments issued, whichever is more reliably measured. Compensation expense for options granted to non-employees was periodically re-measured, as the underlying options vested. Had compensation cost for Gaiam's stock-based compensation plan been determined under the fair value methodology under SFAS No. 123, Gaiam's net income (loss) and income (loss) per share would have been as follows (in thousands, except per share data):

	Three Months Ended June 30,			Six Month June	ıded	
		2005		2004	2005	2004
Net loss as reported	\$	(766)	\$	(2,215)	\$ (650)	\$ (2,544)
Add: Stock-based compensation expense included in reported net income (loss), net of related tax effects				18	_	36
Deduct: Total stock-based compensation expenses determined under fair value based method for all awards, net of related tax				10		30
effects		(65)		(164)	(145)	(329)
Pro forma	\$	(831)	\$	(2,361)	\$ (795)	\$ (2,837)
Net loss per common share						
As reported	\$	(0.05)	\$	(0.15)	\$ (0.04)	\$ (0.17)
Pro forma	\$	(0.06)	\$	(0.16)	\$ (0.05)	\$ (0.19)
Fully diluted net loss per common share						
As reported	\$	(0.05)	\$	(0.15)	\$ (0.04)	\$ (0.17)
Pro forma	\$	(0.06)	\$	(0.16)	\$ (0.05)	\$ (0.19)

2. Stockholders' Equity

During the first quarter of 2005, Gaiam issued a total of 8,221 shares of Class A common stock to Gaiam's independent directors, in lieu of cash compensation, for services rendered in 2004.

8

3. <u>Comprehensive Income (Loss)</u>

Gaiam's comprehensive income (loss) is comprised of net income (loss) and foreign currency translation adjustment, net of income taxes. Comprehensive income (loss), net of related tax effects, was as follows (in thousands):

	Three Months Ended June 30,			 Six Month June	ıded	
		2005		2004	2005	2004
Net income (loss)	\$	(766)	\$	(2,215)	\$ (650)	\$ (2,544)
Foreign currency translation adjustment, net		(263)		(83)	(328)	85
Comprehensive income (loss), net of taxes	\$	(1,029)	\$	(2,298)	\$ (978)	\$ (2,459)

4. <u>Earnings (Loss) per Share</u>

Basic earnings (loss) per share excludes any dilutive effects of options, warrants and dilutive securities. Basic earnings (loss) per share is computed using the weighted average number of common shares outstanding during the period. Diluted earnings per share is computed using the weighted average number of common and common stock equivalent shares outstanding during the period. Common equivalent shares of 121 thousand and 110 thousand are excluded from the computation of diluted earnings per share for the three months ended June 30, 2005 and 2004, respectively, and 108 and 132 thousand common equivalent shares are excluded from the computation for the six months ended June 30, 2005 and 2004, respectively, because their effect is antidilutive.

The following table sets forth the computation of basic and diluted earnings per share (in thousands, except per share data):

	Three Months Ended June 30,				ths Ended ne 30,		
		2005		2004	2004		2003
Numerator for basic and diluted earnings (loss) per share	\$	(766)	\$	(2,215)	\$ (650)	\$	(2,544)
Denominator:							
Weighted average shares for basis earnings (loss) per share		14,820		14,686	14,820		14,650
Effect of dilutive securities:							
Weighted average of common stock, stock options and warrants							_
Denominators for diluted earnings (loss) per share		14,820		14,686	14,820		14,650
Net income (loss) per share - basic	\$	(0.05)	\$	(0.15)	\$ (0.04)	\$	(0.17)
Net income (loss) per share - diluted	\$	(0.05)	\$	(0.15)	\$ (0.04)	\$	(0.17)

5. <u>Segment Information</u>

Gaiam manages its business and aggregates its operational and financial information in accordance with two reportable segments. The direct to consumer segment contains Gaiam's catalog and Internet sales channels, while the business segment comprises the retailer, media and corporate account channels.

9

Although Gaiam is able to track revenue by sales channel, management, allocation of resources and analysis and reporting of expenses is solely on a combined basis, at the reportable segment level.

Contribution margin is defined as net sales, less cost of goods sold and direct expenses. Financial information for Gaiam's business segments was as follows (in thousands):

	Three Months Ended June 30,				nded			
		2005		2004		2005		2004
Net revenue:								
Direct to consumer	\$	11,959	\$	10,291	\$	25,272	\$	22,429
Business		9,747		6,740		22,758		18,377
Consolidated net revenue		21,706		17,031		48,030		40,806
Contribution margin:								
Direct to consumer		(418)		(478)		(252)		(502)
Business		(853)		(3,156)		(732)		(3,461)
Consolidated contribution loss	\$	(1,271)	\$	(3,634)	\$	(984)	\$	(3,963)
Reconciliation of contribution loss to net loss:								
Other income (expense)		(8)		76		108		132
Income tax benefit		(532)		(1,297)		(402)		(1,413)
Minority interest in net (income) loss of consolidated								
subsidiaries, net of tax		(19)		46		(176)		(126)
Net loss	\$	(766)	\$	(2,215)	\$	(650)	\$	(2,544)

6. <u>Subsequent Events</u>

On July 7, 2005, Gaiam issued and sold 2,821,317 shares of unregistered Class A common stock for an aggregate purchase price of approximately \$18.7 million to certain funds advised by Prentice Capital Management, LP.

On July 8, 2005, Gaiam entered into an asset purchase agreement (the "Purchase Agreement") with GT Brands LLC, GT Merchandising & Licensing LLC, Gym Time, LLC, BSBP Productions LLC, and GoodTimes Entertainment LLC (collectively "GoodTimes") pursuant to which Gaiam agreed to purchase substantially all of the assets of GoodTimes relating to GoodTimes' independent home video entertainment business. In addition, Gaiam agreed to assume certain specified liabilities in connection with the acquisition. Under the Purchase Agreement, the purchase price for the assets is \$40 million, subject to certain adjustments relating to changes in working capital and other matters, and subject to approval of the federal bankruptcy court. If consummated, the sale is expected to close by the end of the third quarter 2005.

On July 29, 2005, Gaiam extended its revolving line of credit agreement with Wells Fargo Bank, N.A. to July 31, 2007.

On August 1, 2005, Gaiam issued 60,000 shares of Class A common stock as purchase consideration for a media catalog business.

On August 3, 2005, Gaiam and Jirka Rysavy, Gaiam's Chairman and Chief Executive Officer, entered into an Insurance and Stock Redemption Agreement pursuant to which Gaiam agreed to use the proceeds of any life insurance Gaiam elects to obtain on Mr. Rysavy to acquire Gaiam shares from Mr. Rysavy's estate. The purchase price for any shares will be determined by a national recognized valuation expert.

On August 4, 2005, Gaiam entered into a series of agreements with Revolution Living, LLC; its founder, Steve Case; and Revolution Living's subsidiary, LIME Media. Under the terms of the Transaction Agreement, Revolution Living will acquire 2,500,000 shares of Gaiam Class A common stock at a price of \$8.00 per share, and Gaiam will purchase newly issued preferred equity in LIME at a price of approximately \$7.5 million. On August 4, 2005, Gaiam, Revolution Living, Rysavy and Case also entered into a Shareholders Agreement, and, as a condition to these transactions, Gaiam agreed to enter into four-year employment agreements with Jirka Rysavy and Lynn Powers, Gaiam's President.

Item 2. Management's Discussion and Analysis of Financial Condition and Results of Operations

The following discussion and analysis of Gaiam's financial condition and results of operations should be read in conjunction with the condensed consolidated financial statements and related notes included elsewhere in this document.

10

Overview and Outlook

Gaiam is a lifestyle media company providing content, information, and products to customers who value personal development, healthy living, ecological lifestyles and responsible, inspirational media. Gaiam was incorporated in Boulder, Colorado in 1988.

With the inception of our business segment in 1998, Gaiam started to sell directly to retailers, both domestically and abroad. As of June 30, 2005, Gaiam products were available in over 25,000 stores. Gaiam introduced its "store-within-store" lifestyle presentations, which utilize custom fixtures designed and produced by Gaiam, in late 2000, and the placement of this concept grew to over 4,300 stores by the end of June 2005. In 1998, revenue generated by the business segment was \$3.8 million. In 2004, revenue in this segment was \$44.2 million. During the first six months of 2005, revenue in the business segment was \$22.8 million, an increase of 24% over the first six months of 2004. Gaiam will continue to focus on the national retailer and media channels and plans to continue to build this segment through a combination of new media releases, store within store concept, and expansions in mass market retailers, grocery, bookstores and other media outlets.

The direct to consumer segment continues to be an integral part of Gaiam's outreach. This segment accounted for 54%, or approximately \$52.4 million, of Gaiam's revenue during 2004. During the first six months of 2005, revenue in this segment was \$25.3 million, an increase of 13% over 2004, and represented 53% of total revenue. Through its diverse media reach, the direct to consumer segment provides branding, an opportunity to launch and support new media releases, a sounding board for new product testing, promotional opportunities and customer feedback on Gaiam's and the LOHAS (Lifestyles of Health and Sustainability) industry's focus and future.

In June 2005, Gaiam entered into a stock purchase agreement with certain funds advised by Prentice Capital Management, LP in which Gaiam agreed to sell 2,821,317 shares of unregistered Class A common stock for an aggregate purchase price of \$18.7 million. This transaction closed on July 7, 2005.

On July 8, 2005, Gaiam entered into a definitive agreement to acquire substantially all of the assets of various companies operating as GoodTimes Entertainment. GoodTimes Entertainment is a media company that creates and distributes entertainment programming and home video products through various channels, including television, theaters, retailers and the Internet. GoodTimes Entertainment's library contains approximately 2,000 titles, including wellness franchises such as The Firm and Tae Bo, children classics such as Benji and numerous theatrical releases. The agreement is subject to a number of conditions, including approval of the federal bankruptcy court.

If the acquisition of GoodTimes Entertainment is consummated, Gaiam expects to generate over \$200 million in annual revenues and, in addition to its television and theatrical distribution, Gaiam's home media will be carried by over 40,000 retail stores in U.S., which will be the widest direct distribution coverage in the country for any content provider. Gaiam also expects to distribute 20 million catalogs per year and have approximately 7 million direct customers.

Gaiam's agreement to acquire substantially all of GoodTimes Entertainment's assets for \$40 million in cash (subject to certain adjustments) and assumption of certain liabilities is expected to close the transaction by the end of the third quarter. The acquisition will be financed by existing cash and, if necessary, borrowings under Gaiam's credit facility.

Results of Operations

The following table sets forth certain financial data as a percentage of revenue for the periods indicated:

	Three Month June 3		Six Months I June 30	
	2005	2004	2005	2004
Net revenue	100.0%	100.0%	100.0%	100.0%
Cost of goods sold	51.4%	49.9%	49.4%	48.0%
Gross profit	48.6%	50.1%	50.6%	52.0%

Eumangagu				
Expenses:				
Selling and operating	45.8%	59.7%	45.0%	51.8%
Corporate, general and administration	8.7%	11.7%	7.6%	9.9%
Total expenses	54.5%	71.4%	52.6%	61.7%
Loss from operations	-5.9%	-21.3%	-2.0%	-9.7%
Other income	0.0%	0.4%	0.2%	0.3%
Income tax benefit	-2.5%	-7.6%	-0.8%	-3.5%
Minority intreest in net (income) loss of consolidated				
subsidiaries, net of tax	-0.1%	0.3%	-0.4%	-0.3%
Net loss	-3.5%	-13.0%	-1.4%	-6.2 %

Three months ended June 30, 2005 versus June 30, 2004

Revenue of \$21.7 million for the three months ended June 30, 2005 increased 27.4% from \$17 million for the three months ended June 30, 2004. Gaiam's increase in revenue was primarily due to a \$3 million, or 45%, increase in revenue generated by the business segment, largely as a result of increased volume generated in the mass market segment, additional media placements in Target, and lower product returns. Direct to consumer segment revenue also increased by 16%, fueled by continued strong growth in revenue generated by our Internet business.

Gross profit, which consists of revenue less cost of sales (primarily merchandise acquisition costs and in-bound freight), increased to \$10.6 million for the second quarter of 2005 from \$8.5 million during the same period in 2004. As a percentage of revenue, gross profit of 48.6% for the second quarter of 2005 was approximately the same as that generated during the fourth quarter of 2004, but declined from the 50.1% gross profit generated in the second quarter of 2004. As compared to second quarter of 2004, the lower 2005 gross profit percentage was primarily due to a sales/product mix shift, increased promotional programs in the direct segment during the 2005 period, and higher importation and inbound freight costs associated in part with the rise in fuel costs.

Selling and operating expenses (which consist primarily of sales and marketing costs, commission and fulfillment expenses) decreased to \$9.9 million for the three months ended June 30, 2005 as compared to \$10.2 million in the same period of 2004, even with an overall increase in the sales, due to lower outbound fulfillment costs and bad debt expense. As a percentage of revenue, selling and operating expenses decreased to 45.8% in 2005 from 59.7% in 2004 as a result of the increase in the sales base.

Corporate, general and administration expenses decreased to \$1.9 million for the second quarter of 2005 compared to \$2 million for the same 2004 period, primarily due to expense reductions in our Broomfield headquarters. As a percentage of revenue, corporate, general and administration expenses decreased to 8.7% in the second quarter of 2005 from 11.7% in the comparable 2004 period, as a result of the aforementioned expense reductions and an increase in the sales base.

The operating loss was \$1.3 million for the three months ended June 30, 2005 compared to an operating loss of \$3.6 million for the three months ended June 30, 2004, primarily due to a revenue increase in the second quarter of 2005 as compared to the same period during 2004.

Gaiam recorded \$8 thousand in other expense and \$76 thousand in other income for the three months ended June 30, 2005 and 2004, respectively. The share of net income associated with minority interest was \$19 thousand during the second quarter of 2005, compared to the share of net loss of \$46 thousand for the comparable 2004 period.

Gaiam recorded an income tax benefit of \$532 thousand for the three months ended June 30, 2005 compared to an income tax benefit of \$1.3 million for the three months ended June 30, 2004. Gaiam's consolidated

12

effective tax rate fluctuates based upon the distribution of earnings/losses between its domestic and foreign operations.

As a result of the factors described above, Gaiam's net loss was \$766 thousand, or \$0.05 per share, for the three months ended June 30, 2005, as compared to a net loss of \$2.2 million, or \$0.15 per share, for the three months ended June 30, 2004.

Gaiam generated cash from operating activities of \$2.2 million for the three months ended June 30, 2005, and ended the quarter with \$12.3 million in cash, up from \$10.9 million at March 31, 2005.

Six months ended June 30, 2005 versus June 30, 2004

Revenues of \$48 million for the six months ended June 30, 2005 increased 17.7% from \$40.8 million for the six months ended June 30, 2004. Gaiam's increase in revenues was due primarily to an increase in the business segment of \$4.4 million, or 24% for the first six months of 2005. This increase resulted primarily from higher sales volumes with mass market retailers, coupled with lower product returns. Direct to consumer segment revenues were \$25.3 million for the six months ended June 30, 2005, compared to \$22.4 million for the same period in 2004. The internal growth rate for this segment was 14% for the first six months of 2005, largely due to increases in Internet performance.

Gross profit increased to \$24.3 million for the first six months of 2005 from \$21.2 million during the same period in 2004. As a percentage of revenue, gross profit declined from 52% in 2004 to 50.6% in 2005. This was primarily attributable to a decline in the direct to consumer segment gross margin associated with a change in sales mix to divisions with lower gross margins and more promotional programs during the first six months of 2005. Gross margin in the business segment improved slightly during the first six months of 2005

as compared to the same 2004 period.

Selling and operating expenses increased to \$21.6 million for the six months ended June 30, 2005 as compared to \$21.1 million in the same period of 2004, primarily resulting from increased sales and marketing efforts in our business segment, both domestically and in our U.K. subsidiary, and in our direct to consumer segment to support the revenue increases described above. As a percentage of revenues, selling and operating expenses decreased to 45% in 2005 from 51.8% in 2004.

Corporate, general and administration expenses decreased to \$3.7 million for the six months ended June 30, 2005 from \$4.1 million in the comparable 2004 period, and decreased to 7.6% of revenues in 2005 from 9.9% of revenues in 2004. This expense reduction was generally attributable to cost savings generated by expense reductions in Gaiam's Colorado headquarters.

Operating losses decreased to \$650 thousand for the six months ended June 30, 2005 from \$2.5 million for the six months ended June 30, 2004, primarily due to the increase in revenue and expense reductions.

Gaiam recorded \$108 thousand and \$132 thousand in other income, primarily from interest income, for the six months ended June 30, 2005 and 2004, respectively. The share of net income associated with minority interest was \$176 thousand for the first six months of 2005, compared to the share of net income of \$126 thousand for the comparable 2004 period. This change was primarily the result of higher 2005 minority interest expense associated with the U.K. subsidiary as compared to 2004.

Gaiam recorded an income tax benefit of \$402 thousand for the six months ended June 30, 2005 compared to an income tax benefit of \$1.4 million for the six months ended June 30, 2004.

As a result of the factors described above, Gaiam's net loss was \$650 thousand for the six months ended June 30, 2005 compared to a net loss of \$2.5 million recorded for the six months ended June 30, 2004.

Liquidity and Capital Resources

Gaiam's capital needs arise from working capital required to fund our operations, capital expenditures related to acquisition and development of media content, development of Internet and new products, acquisitions of

13

new businesses, replacements, expansions and improvements to Gaiam's infrastructure, and future growth. These capital requirements depend on numerous factors, including the rate of market acceptance of Gaiam's product offerings, our ability to expand Gaiam's customer base, the cost of ongoing upgrades to Gaiam's product offerings, our level of expenditures for sales and marketing, our level of investment in distribution systems and facilities and other factors. The timing and amount of these capital requirements are variable and cannot accurately be predicted. Additionally, Gaiam will continue to pursue opportunities to expand our media libraries, evaluate possible investments in businesses, products and technologies, and increase our sales and marketing programs and brand promotions as needed.

Gaiam has a credit agreement with Wells Fargo, which permits borrowings of up to \$15 million based upon the collateral value of Gaiam's accounts receivable and inventory. At June 30, 2005, Gaiam had no amounts outstanding under this agreement and complied with all of the financial covenants. This credit agreement expires on July 31, 2007. Should Gaiam choose to borrow under the credit agreement, outstanding advances bear interest at the lower of prime rate less 50 basis points or LIBO plus 275 basis points. Borrowings are secured by a pledge of Gaiam's assets, and the agreement contains various financial covenants, including covenants prohibiting the payment of cash dividends to Gaiam shareholders and requiring compliance with certain financial ratios.

Gaiam's operating activities generated net cash of \$2.8 million and \$4.4 million for the six months ended June 30, 2005 and 2004, respectively. Gaiam's net cash generated from operating activities for both periods is primarily attributable to a decrease in accounts receivable of \$5.6 million in 2005 and \$9.3 million in 2004, and non-cash charges to net income of \$2.5 million in 2005 and \$2.7 million in 2004. These sources of funds were partially offset by the use of funds to reduce accounts payable \$3.9 million in 2005 and \$4.6 million in 2004.

Gaiam's investing and acquisition activities used net cash of \$767 thousand and \$651 thousand during the first six months of 2005 and 2004, respectively, for the purchase of property, equipment and media rights.

Gaiam's financing activities used net cash of \$37 thousand for the six months ended June 30, 2004, primarily to fund principal payments under capitalized leases, and provided net cash of \$402 thousand from the exercise of stock options.

On July 7, 2005, we issued and sold 2,821,317 shares of unregistered Class A common stock for an aggregate purchase price of approximately \$18.7 million to certain funds advised by Prentice Capital Management, LP.

On July 8, 2005, we entered into a purchase agreement with GoodTimes pursuant to which we agreed to purchase substantially all of the assets of GoodTimes relating to GoodTimes' independent home video entertainment business. In addition, we agreed to assume certain specified liabilities in connection with the acquisition. Under the purchase agreement, the purchase price for the assets is \$40 million, subject to certain adjustments relating to changes in working capital and other matters, and subject to approval of the federal bankruptcy court. If consummated, the sale is expected to close by the end of the third quarter 2005. The acquisition will be financed by existing cash and, if necessary, borrowings under our credit facility.

We believe our available cash, cash expected to be generated from operations, cash generated by the sale of unregistered Class A common stock, and borrowing capabilities on our unused \$15 million line of credit, should be sufficient to fund our operations on both a

short-term and long-term basis. However, our projected cash needs may change as a result of acquisitions, product development, unforeseen operational difficulties or other factors.

In the normal course of our business, we investigate, evaluate and discuss acquisition, joint venture, minority investment, strategic relationship and other business combination opportunities in the LOHAS market. For any future investment, acquisition or joint venture opportunities, we may consider using then-available liquidity, issuing equity securities or incurring additional indebtedness.

Contractual Obligations

Gaiam has commitments pursuant to lease and debt agreements, but does not have any outstanding commitments pursuant to long-term debt obligations, capital lease obligations or purchase obligations. The following table shows our commitments to make future payments under operating leases (in thousands):

14

	Total	<	1 year	1-	3 years	3-	5 years	>	5 yrs
Operating lease obligations	\$ 3,762	\$	1,495	\$	1,606	\$	322	\$	339

Off-Balance Sheet Arrangements

Gaiam does not participate in transactions that generate relationships with unconsolidated entities or financial partnerships, such as special purpose entities ("SPEs") or variable interest entities ("VIEs"), which have been established for the purpose of facilitating off-balance sheet arrangements or other limited purposes. As of June 30, 2005, Gaiam is not involved in any unconsolidated SPEs or VIEs.

Critical Accounting Policies

The consolidated financial statements are prepared in conformity with accounting principles generally accepted in the United States, which require management to make judgments, estimates and assumptions that affect the amounts reported in Gaiam's consolidated financial statements and accompanying notes. Note 1 to the consolidated financial statements in Item 8 of our Annual Report on Form 10-K summarizes the significant accounting policies and methods used in the preparation of Gaiam's consolidated financial statements. No changes were made to our significant policies during the three months ended June 30, 2005.

Management believes the following to be critical accounting policies whose application has a material impact on Gaiam's financial presentation, and involve a higher degree of complexity, as they require management to make judgments and estimates about matters that are inherently uncertain.

Provisions for Doubtful Accounts and Returns

Gaiam records a provision for doubtful accounts for all receivables not expected to be collected. Gaiam generally does not require collateral. Gaiam evaluates the collectibility of accounts receivable based on a combination of factors. In circumstances in which we are aware of a specific customer's inability to meet its financial obligations (e.g. bankruptcy filings), Gaiam records a specific reserve for bad debts against amounts due. For all other instances, Gaiam recognizes reserves based on historical experience and review of individual accounts outstanding.

Gaiam records a provision for product returns to be received in future periods at the time the original sale is recognized. The amount of the returns provision is based upon historical experience and future expectations.

Inventory

Inventory consists primarily of finished goods held for sale and is stated at the lower of cost (first-in, first-out method) or market. Gaiam identifies the inventory items to be written down for obsolescence based on the item's current sales status and condition. If the item is discontinued or slow moving, it is written down based on an estimate of the markdown to retail price needed to sell through its current stock level of the item.

Goodwill

Goodwill represents the excess of the purchase consideration over the fair value of assets acquired less liabilities assumed in a business acquisition. In accordance with SFAS No. 142, "Goodwill and Other Intangible Assets," goodwill is no longer amortized but is reviewed for impairment annually, or more frequently if impairment indicators arise, on a reporting unit level. The fair value of a reporting unit is compared with its carrying amount, including goodwill. If the fair value of a reporting unit exceeds its carrying amount, goodwill of the reporting unit is considered not impaired.

If the carrying amount of a reporting unit exceeds its fair value, the goodwill impairment test is performed to measure the amount of impairment loss. Gaiam has allocated goodwill to two reporting units, and uses a market value method for the purposes of testing for potential impairment. The annual review requires extensive use of financial judgment and estimates. Application of alternative assumptions and definitions, such as a change in the composition of a reporting unit, could yield significantly different results.

Investments

Investments in entities over which Gaiam does not have the ability to exercise significant influence are accounted for under the cost method. Under the cost method of accounting, investments in private companies are carried at cost and are adjusted only for other-than-temporary declines in fair value. Investments under the cost method are included on the accompanying consolidated balance sheet in "Investments." Had Gaiam concluded that it had the ability to exercise significant influence, its share of the investee's income or loss would have been reported in Gaiam's consolidated statement of operations.

Capitalized Production Costs

Capitalized production costs include costs incurred to produce informational media products marketed by Gaiam to retail marketers and direct-mail and online customers. These costs are deferred for financial reporting purposes until the media is released, then amortized over succeeding periods on the basis of estimated sales. Historical sales statistics are the principal factor used in estimating the amortization rate.

Gaiam had no significant changes in its critical accounting policies from its Form 10-K filing for the fiscal year ended December 31, 2004.

Risk Factors

We wish to caution you that there are risks and uncertainties that could cause our actual results to be materially different from those indicated by forward-looking statements that we make from time to time in filings with the Securities and Exchange Commission, news releases, proxy statements, annual reports, registration statements and other written communications, as well as oral forward-looking statements made from time to time by representatives of Gaiam. These risks and uncertainties include, but are not limited to, those listed in Gaiam's Annual Report on Form 10-K for the year ended December 31, 2004. These risks and uncertainties and additional risks and uncertainties not presently known to us or that we currently deem immaterial may cause our business, financial condition, operating results and cash flows to be materially adversely affected. Except for the historical information contained herein, the matters discussed in this analysis are forward-looking statements that involve risk and uncertainties, including, but not limited to, general economic and business conditions, competition, pricing, brand reputation, consumer trends, and other factors which are often beyond our control. Gaiam does not undertake any obligation to update forward-looking statements except as required by law.

Item 3. Quantitative and Qualitative Disclosures About Market Risk

We are exposed to market risks, which include changes in U.S. interest rates and foreign exchange rates. We do not engage in financial transactions for trading or speculative purposes.

Any borrowings we might make under our bank credit facility would bear interest at the lower of prime rate less 50 basis points or LIBOR plus 275 basis points. Gaiam does not have any amounts outstanding under its credit facility, so any unfavorable change in interest rates would not have a material impact on Gaiam's results from operations or cash flows unless Gaiam makes borrowings in the future.

Gaiam purchases a significant amount of inventory from vendors outside of the U.S. in transactions that are primarily U.S. dollar denominated transactions. Because the percentage of our international purchases denominated in currencies other than the U.S. dollar is small, any currency risks related to these transactions are immaterial to Gaiam. A decline in the relative value of the U.S. dollar to other foreign currencies could, however, lead to increased purchasing costs. In order to mitigate this exposure, Gaiam makes virtually all of its purchase commitments in U.S. dollars.

In 2003, Gaiam purchased a 50.1% interest in Leisure Systems International Limited, a U.K. based distributor. Because Leisure Systems' revenue is primarily denominated in foreign currencies, this investment exposes Gaiam to accounting risk associated with foreign currency exchange rate fluctuations. However, we have determined that no material market risk exposure to our consolidated financial position, results of operations or cash flows existed as of June 30, 2005.

16

Item 4. Controls and Procedures

Evaluation of Disclosure Controls and Procedures

Gaiam's chief executive officer and chief financial officer conducted an evaluation of the effectiveness of the design and operation of Gaiam's disclosure controls and procedures, as defined in Rules 13a-15(e) and Rule 15d-15(e) under the Exchange Act. Based upon their evaluation as of June 30, 2005, Gaiam's chief executive officer and chief financial officer have concluded that those disclosure controls and procedures are effective.

Changes in Internal Control over Financial Reporting

No changes in Gaiam's internal control over financial reporting occurred during the quarter ended June 30, 2005 that have materially affected, or are reasonably likely to materially affect, Gaiam's internal control over financial reporting.

PART II. OTHER INFORMATION

Item 1. Legal Proceedings

From time to time, Gaiam is involved in legal proceedings that we consider to be in the normal course of business. We do not believe that any of these proceedings will have a material adverse effect on our business.

Item 2. Sales of Unregistered Securities and Use of Proceeds

On August 1, 2005, Gaiam issued 60,000 shares of unregistered Class A common stock, valued at \$9.20 per share, to Conscious Wave, Inc., as purchase consideration for a media catalog business. These shares were issued pursuant to an exemption from registration under Section 4(2) of the Securities Act of 1933.

Item 3. Defaults Upon Senior Securities

None.

Item 4. Submission of Matters to a Vote of Security Holders.

On May 19, 2005, Gaiam held its Annual Meeting of Shareholders. The shareholders elected seven directors to serve until the next annual meeting of shareholders to be held in 2006 or until their successors are duly elected and qualified. The results of this vote follow:

Jirka Rysavy	For:	60,970,492	Withheld:	1,469,655
Lynn Powers	For:	60,970,511	Withheld:	1,469,636
James Argyropoulos	For:	62,384,610	Withheld:	55,537
Barnet Feinblum	For:	62,362,929	Withheld:	77,218
Barbara Mowry	For:	62,384,594	Withheld:	55,553
Ted Nark	For:	62,384,478	Withheld:	55,669
Paul Ray	For:	62,132,802	Withheld:	307,345

Item 5. Other Information

None.

Item 6. Exhibits

a) Exhibits

Exhibit No.	Description
10.1	Form of Stock Option Agreement under Gaiam's 1999 Long-Term Incentive Plan.
10.2	Amended and Restated Credit Agreement dated July 29, 2005 between Gaiam, Inc. and
	Wells Fargo Bank, National Association.
	17
31.1	Certification of the Chief Executive Officer pursuant to Rule 13a-14(a) under the Securities Exchange Act of 1934 (filed herewith).
31.2	Certification of the Chief Financial Officer pursuant to Rule 13a-14(a) under the Securities Exchange Act of 1934 (filed herewith).
32.1	Certification of the Chief Executive Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (furnished herewith).
32.2	Certification of the Chief Financial Officer pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (furnished herewith).
	18

Signatures

In accordance with the requirements of the Securities and Exchange Act, the registrant caused this report to be signed on its behalf, by the undersigned, thereunto duly authorized.

Gaiam, Inc. (Registrant) August 9, 2005

By: /s/ Jirka Rysavy

Jirka Rysavy

Chief Executive Officer

By: /s/ Janet Mathews
Janet Mathews
Chief Financial Officer

GAIAM, INC.

Employee Stock Option Agreement

This Stock Option Agreement set forth below (this "Agreement") is dated as of the date of grant set forth below and is between Gaiam, Inc., a Colorado corporation ("Gaiam"), and the above named individual (the "Grantee").

Gaiam has established its 1999 Long-Term Incentive Plan (the "Plan") to advance the interests of Gaiam and its shareholders by providing incentives to certain eligible persons who contribute significantly to the strategic and long-term performance objectives and growth of Gaiam and any parent or subsidiary of Gaiam.

This Agreement evidences an option grant as follows:

Granted to: {Employee's name}
Number of Shares: {XXX,XXX}
Effective Date of Grant: {Month, Day, Year}
Expiration Date: 7 years from Date of Grant

Exercise Price Per Share: \$XX.XX

Vesting Dates: Two percent (2%) of the shares shall vest on each vesting date.

The first vesting date shall be {{Month, Date, Year} equal to the 1st day of the eleventh month after the grant date}, and a vesting date shall occur on the first day of each month thereafter until {Month, Day, Year}, when all shares shall be vested. Vesting stops if Grantee ceases to be employed by the

Company.

Pursuant to the provisions of the Plan, the Board of Directors of Gaiam (the "Board") or a Committee designated by the Board (the "Committee") has full power and authority to direct the execution and delivery of this Agreement in the name and on behalf of Gaiam. The Board or the Committee authorized the execution and delivery of this Agreement. All capitalized terms not otherwise defined in this Agreement have the same meaning given such capitalized terms in the Plan.

Agreement

The parties agree as follows:

Section 1. Grant of Stock Option; Term. Subject and pursuant to all terms and conditions stated in this Agreement and in the Plan, Gaiam hereby grants to Grantee an option (the "Option") to purchase the number of shares (the "Option Shares") of Gaiam's Class A common stock, par value \$0.0001 per share (the "Common Shares"), set forth above, at the exercise price set forth above. Except as otherwise provided in this Agreement or the Plan, the Option may not be exercised after the close of business on the expiration date set forth above. Grantee hereby accepts the Option on such terms and conditions, including, without limitation, the confidentiality and non-compete provisions set forth in Section 8 of this Agreement. The Option is a Nonqualified Stock Option (as such term is defined in the Plan). Grantee shall, subject to the limitations of this Agreement and the Plan, have the right to exercise the Option by purchasing all or any part of the vested Option Shares then available for purchase under the vesting schedule set forth above (less any Option Shares previously purchased upon exercise of this Option).

Section 2. *Procedures for Exercise*. Grantee shall exercise all or any part of the Option by delivering to Gaiam: (i) written notice of the number of vested Option Shares to be purchased, (ii) a duly

executed Stock Restriction Agreement substantially in the form of Appendix A (unless such agreement would have expired according to its terms), (iii) payment of the exercise price of such Option Shares in the form of cash or, if permitted by the Committee, qualified Common Shares, the surrender of another outstanding Award under the Plan or any combination thereof, and (iv) payment of any required withholding pursuant to Section 10. The Option shall be deemed to have been exercised as of the close of business on the date the required documents and required consideration are received by Gaiam. For purposes of this Section 2, Common Shares shall be deemed to be "qualified" Common Shares if they have been held by Grantee for six months or such other period as set from time to time by the Board or the Committee.

Section 3. Termination of Employment, Retirement, Disability or Death.

- (a) Vesting shall cease on the date Grantee ceases to be employed by the Company. Following Grantee's last day of employment with the Company, this Option shall only be exercisable for the number of Option Shares that are vested as of Grantee's last day of employment with by the Company (less any Option Shares previously acquired upon exercise of this Option).
 - (b) Except as provided in Section 3(c) or 3(d), following Grantee's last day of employment with the Company, this Option may be exercised at any time and from time to time within the lesser of (i) the 30 day period commencing on the first day after Grantee's last day of employment with the Company or (ii) the remaining term of the Option.
 - (c) If termination of employment occurs due to death or disability while Grantee is an employee of the Company,

then this Option may be exercised at any time and from time to time within the lesser of (i) the one year period commencing on the first day after Grantee's last day of employment with the Company or (ii) the remaining term of the Option.

- (d) If termination of employment occurs due to retirement at or after normal retirement age, as prescribed from time to time by the Company's retirement policy, or retirement under circumstances approved by the Committee (either before or after retirement), then this Option may be exercised at any time within the lesser of (i) the three month period commencing on the first day after Grantee's last day of employment with the Company, or, if Grantee dies during the three month period commencing on the first day after Grantee's last day of employment with the Company, then the one year period commencing on the first day after Grantee's last day of employment with the Company, or (ii) the remaining term of the Option.
- **Section 4.** *Issuance and Delivery of Option Shares*. The stock certificate(s) representing Option Shares shall be issued to Grantee subject to satisfaction of the applicable tax withholding requirements set forth in Section 10. The issuance of Option Shares shall be in accordance with the provisions of Section 5.
- **Section 5.** *No Issuance of Option Shares if Violation*. Gaiam shall not issue stock certificate(s) representing Option Shares if the administrator of the Plan or its authorized agent determines, in its sole discretion that the issuance of such certificate would violate the terms of the Plan, this Agreement or applicable law.
- Section 6. *Rights as an Employee or Shareholder*. Except as otherwise provided in the Plan, no person shall be, or have any of the rights or privileges of, a shareholder of Gaiam with respect to any of the Option Shares unless and until certificates representing such shares shall have been issued and delivered to such person. Neither the Plan nor this Agreement shall be deemed to give Grantee any right with respect to continued employment with the Company, nor shall the Plan or the Agreement be deemed to limit in any way the Company's right to terminate Grantee's employment at any time.
- **Section 7.** *Nondisparagement and Further Assistance.* During Grantee's employment and thereafter, Grantee will not make any disclosure, issue any public statements or otherwise cause to be disclosed any information which is designed, intended or might reasonably be anticipated to discourage

suppliers, customers or employees of the Company or otherwise have a negative impact or adverse effect on the Company. Grantee will provide assistance reasonably requested by the Company in connection with actions taken by Grantee while employed by the Company, including but not limited to assistance in connection with any lawsuits or other claims against the Company arising from events during the period in which Grantee was employed.

Section 8. Nondisclosure of Confidential Information; Non-Compete. In consideration of the receipt of the Option, Grantee agrees (i) not to disclose to any third party any trade secrets or any other confidential information of the Company (including but not limited to cost or pricing information, customer lists, commission plans, supply information, internal business procedures, market studies, expansion plans, potential acquisitions, terms of any acquisition or potential acquisition or the existence of any negotiations concerning the same or any similar non-public information relating to the Company's internal operations, business policies or practices) acquired during Grantee's employment by the Company or after the termination of such employment, or (ii) use or permit the use of any of the Company's trade secrets or confidential information in any way to compete (directly or indirectly) with the Company or in any other manner adverse to the Company. In addition, Grantee agrees that, without the prior written consent of the Company, signed by the Company's Chief Executive Officer, Grantee will not, during the term of Grantee's employment by the Company or for a period of two years thereafter (i) accept employment with, serve as a consultant to, or accept compensation from any person, firm or corporation (including any new business started by Grantee, either alone or with others) whose products and or services compete with those offered by the Company, in any geographic market in which the Company is then doing business or to Grantee's knowledge plans to do business, (ii) contact or solicit any customers of the Company for the purposes of diverting any existing or future business of such customers to a competing source, (iii) contact or solicit any vendors to the Company (directly or indirectly) for the purpose of causing, inviting or encouraging any such vendor to alter or terminate his, her or its business relationship with the Company, or (iv) contact or solicit any employees of the Company (directly) for the purpose of causing, inviting or encouraging any such employee to alter or terminate his, her or its employment relationship with the Company.

The Company will be entitled to enforce its rights under this Agreement specifically, to recover damages by reason of any breach of any provision of this Agreement and to exercise all other rights to which it may be entitled. Grantee agrees and acknowledges that money damages may not be an adequate remedy for breach of the provisions of this Agreement and that the Company may in its sole discretion apply to any court of law or equity of competent jurisdiction for specific performance and/or injunctive relief in order to enforce or prevent any violations of the provisions of this Agreement.

Grantee agrees that this covenant is reasonable with respect to its duration, geographic area and scope. It is the desire and intent of the parties that the provisions of this Section 8 shall be enforced to the fullest extent permissible under the laws and public policies applied in each jurisdiction in which enforcement is sought. Accordingly, if any particular portion of this Section 8 shall be adjudicated to be invalid or unenforceable, this Section 8 shall be deemed amended to delete therefrom the portion thus adjudicated to be invalid or unenforceable, such deletion to apply only with respect to the operation of this Section 8 in the particular jurisdiction in which such adjudication is made.

Section 9. *Securities Laws*. Grantee acknowledges that applicable securities laws may restrict the right and govern the manner in which Grantee may dispose of the Option Shares obtained upon exercise of the Option and Grantee agrees not to offer, sell or otherwise dispose of any such shares in a manner that would violate the Securities Act of 1933, as amended, or any other federal or state law. Gaiam intends, within 12 months of its initial public offering of shares, to file a registration statement on Form S-8 to register the shares to be issued upon exercise of this Option.

Section 10. *Income Taxes.* Grantee acknowledges that when Grantee is required to recognize income for federal, state or local income tax purposes on account of the grant, vesting and/or exercise of the Option, pursuant to this Agreement, that such income shall be subject to withholding of tax by the Company. Grantee agrees that the Company may either withhold an appropriate amount from any

compensation or any other payment of any kind then payable or that may become payable to Grantee or, require Grantee to make a cash payment to the Company equal to the amount of withholding required in the opinion of the Company. In the event Grantee does not make such payment when requested, the Company may refuse to issue or cause to be delivered any shares under this Agreement or any other incentive plan agreement entered into by Grantee and the Company until such payment has been made or arrangements for such payment satisfactory to the Company have been made. Grantee agrees further to notify the Company promptly if Grantee files an election pursuant to Section 83(b) of the Internal Revenue Code of 1986, as amended, with respect to any Option Shares.

Section 11. *Prohibition on Transfer or Assignment*. Except as provided in the Plan, neither this Agreement nor the Option may be transferred or assigned, other than an assignment by will or by laws of descent and distribution, and this Option shall be exercisable during the Grantee's lifetime only by Grantee or by such permitted assignee.

Section 12. *Binding Effect; No Third Party Beneficiaries*. This Agreement shall be binding upon and inure to the benefit of the Company and Grantee and their respective heirs, representatives, successors and permitted assigns. This Agreement shall not confer any rights or remedies upon any person other than Gaiam and the Grantee and their respective heirs, representatives, successors and permitted assigns. The parties agree that this Agreement shall survive the exercise or termination of the Option.

Section 13. Agreement to Abide by Plan; Conflict between Plan and Agreement. The Plan is hereby incorporated by reference into this Agreement and made a part hereof as though fully set forth in this Agreement. Grantee, by execution of this Agreement, (i) represents that he is familiar with the terms and provisions of the Plan and (ii) agrees to abide by all of the terms and conditions of this Agreement and the Plan. Grantee accepts as binding, conclusive and final all decisions or interpretations of the administrator of the Plan upon any question arising under the Plan and this Agreement (including, without limitation, the cause of any termination of Grantee's employment with the Company). In the event of any conflict between the Plan and this Agreement, the Plan shall control and this Agreement shall be deemed to be modified accordingly.

Section 14. *Entire Agreement.* This Agreement constitutes the entire agreement between the parties and supersedes any prior understandings, agreements, or representations by or between the parties, written or oral, to the extent they related in any way to the subject matter hereof.

Section 15. *Choice of Law.* To the extent not superseded by federal law, the laws of the state of Colorado shall control in all matters relating to this Agreement and any action relating to this Agreement must be brought in Denver, Colorado.

Section 16. Notice. All notices, requests, demands, claims, and other communications under this Agreement shall be in writing. Any notice, request, demand, claim, or other communication under this Agreement shall be deemed duly given if (and then two business days after) it is sent by registered or certified mail, return receipt requested, postage prepaid, and addressed to the intended recipient at the address set forth below the recipient's signature to this Agreement. Either party to this Agreement may send any notice, request, demand, claim, or other communication under this Agreement to the intended recipient at such address using any other means (including personal delivery, expedited courier, messenger service, telecopy, ordinary mail, or electronic mail), but no such notice, request, demand, claim, or other communication shall be deemed to have been duly given unless and until it actually is received by the intended recipient. Either party to this Agreement may change the address to which notices, requests, demands, claims, and other communications hereunder are to be delivered by giving the other party notice in the manner set forth in this section.

Section 17. *Counterparts*. This Agreement may be executed in one or more counterparts, each of which shall be deemed an original but all of which together shall constitute one and the same instrument.

I	EXECUTED as of the date of grant set forth at	pove.	
GAIAM,	INC.	GRANTEE	
By {Exe	ecutive Officer}	{Employee's name}	
Address: 360 Interlocken Blvd. Broomfield, Colorado 80021 Attn.: Stock Option Administration	Broomfield, Colorado 80021	Address:	
	Social Security No.		

2005 AMENDED AND RESTATED

CREDIT AGREEMENT

(REVOLVING LOAN)

BY AND BETWEEN

WELLS FARGO BANK, NATIONAL ASSOCIATION, AS LENDER.

AND

GAIAM, INC., AND THE RESTRICTED SUBSIDIARIES

AS BORROWER

DATED AS OF JULY 29, 2005

CREDIT AGREEMENT (REVOLVING LOAN)

Recitals

- A. WELLS FARGO BANK WEST, N.A. and GAIAM, INC. and its 100% owned subsidiaries, entered into that certain Loan Agreement ("2002 Credit Agreement") dated as of December 31, 2002 ("Original Effective Date").
- B. The parties to the 2002 Credit Agreement desire to make certain amendments to, but not to discharge any indebtedness or other obligations owing under, the 2002 Credit Agreement, as incorporated in this 2005 Amended and Restated Credit Agreement which, upon satisfaction of the provisions in Section 8.1 thereof, shall replace in its entirety the 2002 Credit Agreement.

Agreement

THIS 2005 AMENDED AND RESTATED CREDIT AGREEMENT ("Credit Agreement") is entered into as of July 29, 2005 ("Effective Date"), by and between WELLS FARGO BANK, NATIONAL ASSOCIATION (as successor in interest to Wells Fargo Bank West, N.A.) as lender ("Lender") and GAIAM, INC., a corporation formed under the laws of the State of Colorado, whose address is 360 Interlocken Boulevard, Suite 300, Broomfield, Colorado 80021 ("Gaiam") and the Persons identified on <u>Schedule 1</u> hereto ("Restricted Subsidiaries" and, collectively with Gaiam, the "Borrower").

ARTICLE 1. DEFINED TERMS

As used in this Credit Agreement, the following terms shall have the meanings set forth below (and such meaning shall be equally applicable to both the singular and plural form of the terms defined, as the context may require):

- 1.1 <u>AAA</u>: shall have the meaning set forth in Subsection 13.4.1.
- 1.2 <u>Accounts Aging Report</u>: shall have the meaning set forth in Subsection 9.2.15.
- 1.3 <u>Accounts Receivable</u>: means all rights to payment for goods sold or leased or for services rendered whether or not earned by performance, including those evidenced by an instrument or chattel paper (as those terms are defined in the Colorado Uniform Commercial Code).

- 1.5 Advance Date: a day (which shall be a Banking Day) on which a Revolving Advance is made.
- 1.6 <u>Affiliate</u>: means, as to any Person, any other Person that directly or indirectly controls, or is controlled by, or is under common control with such Person. The term "control" means the possession, directly or indirectly, of the power to direct or cause the direction of the management and policies of a Person, whether through the ownership of voting securities, by contract, or otherwise.
 - 1.7 Aggregate LC Commitment: shall be \$15,000,000.00.
- 1.8 <u>Applicable Lending Office</u>: means for each type of Revolving Advance the lending office of Lender designated as such for such Revolving Advances on its signature page hereof or such other office of Lender as Lender may from time to time specify to Borrower as the office by which its Revolving Advances of such type are to be made and maintained.
 - 1.9 <u>Authorized Officer</u>: shall have the meaning set forth in Subsection 8.1.5.
- 1.10 <u>Bank Debt</u>: all amounts owing under the Revolving Note, fees, and all interest, expenses, charges and other amounts payable by Borrower pursuant to the Loan Documents.
- 1.11 <u>Banking Day</u>: any day (a) other than a Saturday or Sunday and other than a day which is a Federal legal holiday or a legal holiday for banks in the State of Colorado, and (b) if such day relates to a borrowing of, a payment or prepayment of principal of or interest on, a continuation of or conversion into, or a LIBO Rate Period for, a LIBO Rate Loan, or a notice by Borrower with respect to any such borrowing, payment, prepayment, continuation, conversion, or LIBO Rate Period, on which dealings in U.S. Dollar deposits are carried out in the London interbank market.
- 1.12 <u>Base LIBOR</u>: means a rate of interest per annum equal to the rate as published in the Eastern Edition of the *Wall Street Journal* on the Banking Day corresponding to the first day of each LIBO Rate Period (including LIBO Rate Periods that occur on account of the automatic renewal of a pervious LIBO Rate Period as provided in Subsection 4.1.2 hereof) as the Inter-Bank Market Offered Rate for delivery of funds on such date for a 30-day period and in an amount approximately equal to the principal amount to which it would apply, with the understanding that such rate is utilized by Lender for the purpose of calculating effective rates of interest for loans making reference thereto.

- 1.13 <u>Base Rate</u>: means at any time the rate of interest most recently announced by Lender as its Prime Rate, from time to time which serves as the basis upon which effective rates of interest are calculated for those loans making reference thereto, adjusted by the Base Rate Margin.
 - 1.14 Base Rate Loan: shall have the meaning set forth in Subsection 4.1.1.
 - 1.15 <u>Base Rate Margin</u>: negative 50 basis points.
- 1.16 Borrower Benefit Plan: means any "employee benefit plan", as such term is defined in Section 3(3) of ERISA, including the following: (a) any funded "employee welfare benefit plan," as that term is defined in Section 3(1) of ERISA; (b) any "multiemployer plans," as defined in Section 3(37) of ERISA; (c) any "employee pension benefit plan" as defined in Section 3(2) of ERISA; (d) any "multiple employer plan" within the meaning of Section 413 of the Code; (e) any "multiple employer welfare arrangement" within the meaning of Section 3(40) of ERISA; (f) a "voluntary employees" beneficiary association" within the meaning of Section 501(c)(9) of the Code; (g) a "welfare benefit fund" within the meaning of Section 419 of the Code; or (h) any employee welfare benefit plan within the meaning of Section 3(1) of ERISA for the benefit of retired or former employees, which is maintained by the Borrower or in which Borrower participates or to which Borrower is obligated to contribute, but excluding any such plan, arrangement, association or fund that is maintained outside of the United States primarily for the benefit of persons substantially all of whom are nonresident aliens.
- 1.17 <u>Borrower Pension Plan</u>: means each Borrower Benefit Plan that is an "employee pension benefit plan" as defined in Section 3(2) of ERISA that is intended to satisfy the requirements of Section 401(a) of the Code.
- 1.18 <u>Borrower Account</u>: means the deposit accounts for each Borrower maintained at Lender and described on <u>Exhibit 1.18</u> hereto.
- 1.19 <u>Borrowing Base</u>: means the sum of the following amounts determined on any date pursuant to the most recent Borrowing Base Certificate (subject to adjustment based upon any audit done by Lender):
 - (a) Seventy percent (70.0%) of the net unpaid amount of all Eligible Receivables; plus
- (b) Forty percent (40.0%) of the value of all Eligible Inventory, with such value determined at the lower of cost or market in accordance with GAAP.
- 1.20 <u>Borrowing Base Certificate</u>: means the certificate so titled periodically provided by Borrower to Lender and in the form of <u>Exhibit 1.20</u> hereto.

- 1.21 <u>Borrowing Notice</u>: shall have the meaning set forth in Section 2.2.
- 1.22 <u>Capital Expenditures</u>: means an expenditure for the purchase of any fixed asset as determined in accordance with GAAP.
- 1.23 <u>Capital Lease</u>: means any lease of property (whether real, personal or mixed) by a Person, the discounted present value of the rental obligations of such Person as lessee under such lease, in accordance with GAAP, is required to be capitalized on the balance sheet of such Person.
- 1.24 <u>Cash Collateral Account</u>: means an account for the deposit of funds established with Lender, or with such other financial institution as shall be approved by Lender.
- 1.25 <u>Casualty Event</u>: means a loss or taking caused by or resulting from a fire, earthquake, explosion, wind, rain, or condemnation, or substantially similar occurrence.
- 1.26 <u>Casualty Proceeds</u>: the amount received on account of a Casualty Event from insurance, condemnation award, judgment, or settlement.
 - 1.27 <u>Change in Law</u>: shall have the meaning set forth in Subsection 4.2.2.
- 1.28 <u>Closing Date</u>: that date, which must occur on or before July 29, 2005, on which Lender and Borrower have executed all Loan Documents to which they are parties and on which all of the conditions set forth in Section 8.1 of this Credit Agreement have been met.
 - 1.29 Code: means the Internal Revenue Code of 1986, as amended.
 - 1.30 <u>Collateral</u>: shall have the meaning set forth in Section 6.1.
- 1.31 <u>Committed Advances</u>: means the principal amount of all Revolving Advances which Lender is obligated to make as a result of Borrower having made a Borrowing Notice pursuant to Section 2.3 hereof, but which has not been funded.
- 1.32 <u>Compliance Certificate</u>: a certificate of the chief financial officer of Borrower acceptable to Lender and in the form attached hereto as Exhibit 1.32.
- 1.33 <u>Consolidated Subsidiaries</u>: means, considered together, (a) each of Gaiam's Subsidiaries and (b) the following entities in which Gaiam has an equity ownership interest in excess of fifty percent (50.0%): (i) Natural Habitat, Inc., and (ii) Gaiam Limited.

- 1.34 Conversion Request: shall have the meaning set forth in Subsection 4.1.2.
- 1.35 <u>Corporate Officer</u>: shall have the meaning set forth in Section 1.70.
- 1.36 <u>Current Assets</u>: means with respect to any Person, current assets less receivables and investments in or other amounts due from any shareholder, director, officer, employee or any person or entity related to, or an Affiliate of, such Person, determined in accordance with GAAP.
- 1.37 <u>Current Liabilities</u>: means with respect to any Person the aggregate amount of such Person's items properly shown as current liabilities on its balance sheet, determined in accordance with GAAP.
 - 1.38 <u>Default Interest Rate</u>: a rate of interest equal to 400 basis points in excess of the Base Rate in effect from time to time.
 - 1.39 <u>Delinquent Accounts</u>: means Accounts Receivable unpaid sixty (60) days after the due date therefore.
- 1.40 <u>Eligible Inventory</u>: means all of the Inventory of Borrower in which Lender has a first lien security interest (subject only to liens permitted under this Credit Agreement in Section 10.2 or otherwise herein), and which meets the following requirements:
 (a) is not located at premises outside of the United States; (b) does not constitute bill and hold goods except to the extent that the Account Receivable arising from the sale of such goods does not constitute an Eligible Receivable; (c) is currently listed for sale in Borrower's current catalogs or websites, will be listed for sale in future catalogs or on the website, and does not constitute unserviceable, obsolete or slow moving Inventory; (d) does not constitute returned, damaged and/or defective Inventory (including, but not limited to, VHS tapes returned by customers for refund, replacement by DVDs, or otherwise); and (e) is not Inventory purchased on consignment. Inventory which is not Eligible Inventory shall nevertheless be part of the Collateral.
- 1.41 <u>Eligible Receivables</u>: means all of the Accounts Receivable of Borrower in which Lender has a first lien security interest (subject only to liens permitted under this Credit Agreement in Section 10.2 or otherwise herein), and which meet the following requirements: (i) arise from the actual <u>bona fide</u> sale, in the ordinary course of business, of Inventory on ordinary trade terms and in accordance with the terms and provisions contained in any documents related thereto; (ii) are evidenced by an invoice; and (iii) are net of any credit, trade or other allowance given to the account debtor thereof. "Eligible Receivables" shall not include any Accounts Receivable

- (a) The Accounts Receivable of any account debtor twenty-five percent (25%) or more of whose Accounts Receivable are Delinquent Accounts;
 - (b) Which are Delinquent Accounts;
 - (c) With respect to which the account debtor is an Affiliate of Borrower or an employee of Borrower;
- (d) With respect to which the account debtor has asserted in writing any defense, counterclaim, or right to discount, whether well founded or otherwise, or which is subject to liens, encumbrances, or rights in favor of Persons other than Lender (other than liens permitted under this Credit Agreement in Section 10.2 or otherwise), or which are subject to any offset or deduction;
- (e) With respect to which the account debtor is the subject of dissolution, liquidation or termination proceedings, or with respect to which there has been commenced a voluntary or involuntary proceeding under any provision of the bankruptcy or insolvency laws or with respect to which there has been an assignment for the benefit of creditors;
- (f) With respect to which payment by the account debtor may be conditional pursuant to the terms of the engagement;
- (g) With respect to which the account debtor is not a resident of, or have its chief executive office located in, the United States, except to the extent such accounts are supported by (i) an irrevocable letter of credit (the original of which is delivered to Lender or a representative of Lender) issued or confirmed by a bank chartered under the laws of the United States or any state and which has combined capital, surplus and undivided profits of at least \$250,000,000.00, or (ii) insurance, bonds or other assurances reasonably satisfactory to Lender; provided that Accounts Receivable of Borrower which otherwise qualify as Eligible Receivables may be treated as Eligible Receivables up to a maximum of \$500,000.00 in the aggregate even though the chief executive office of the account debtor or debtors thereof is located in Canada.
- (h) With respect to which Borrower is or may become liable to the account debtor for goods sold or services rendered by the account debtor to Borrower; provided that this clause (h) shall not apply to Accounts Receivable where the account debtor is Whole Foods Market, Inc.;
- (i) With respect to which goods have not been shipped, or the services have not been rendered, to the account debtor;
- (j) That portion of any Accounts Receivable which represent interim or progress billings or retention rights on the part of the account debtor thereunder;

- (k) The Accounts Receivable of the account debtor thereunder, and its Affiliates, constitute more than fifty percent (50%) of all otherwise Eligible Receivables (but the portion of the Accounts Receivable of such account debtor not in excess of such percentage may be deemed Eligible Receivables); or
- (l) With respect to which the account debtor is a federal, state, or local government or any department or agency thereof and the aggregate amount of Accounts Receivable owed by such account debtor is in excess of \$10,000.00.

Accounts Receivable which are not Eligible Receivables shall nevertheless be part of the Collateral.

- 1.42 Environmental Laws: means any federal or state law relating to the public health, safety, industrial hygiene, pollution or the environment, including laws relating to noise or to emissions, discharges, releases or threatened releases of Hazardous Substances into the workplace, the community or the environmental conditions, or otherwise relating to the generation, manufacture, processing, distribution, use, treatment, storage, disposal, transport or handling of Hazardous Substances, and including, without limitation, the Comprehensive environmental Response Compensation and Liability Act of 1980 as amended, 42 U.S.C. 9601-9657 ("CERCLA"), the Resource Conservation and Recovery Act of 1976, 42 U.S.C. 6901-6987 ("RCRA"), the Superfund Amendments and Reauthorization Act ("SARA"), and the Federal Toxic Substances Control Act, as each may be amended from time to time.
 - 1.43 <u>Environmental Regulations</u>: as defined in the definition of Hazardous Substances.
 - 1.44 <u>ERISA</u>: shall have the meaning set forth in Section 7.10.
- 1.45 <u>ERISA Affiliate</u>: means any corporation or trade or business which is a member of the same controlled group of corporations (within the meaning of Section 414(b) of the Code) as Borrower or is under common control (within the meaning of Section 414(c) of the Code) with Borrower, provided, however, that for purposes of provisions herein concerning minimum funding obligations (imposed under Section 412 of the Code or Section 302 of ERISA), the term "ERISA Affiliate" shall also include any entity required to be aggregated with Borrower under Section 414(m) or 414(o) of the Code.

- 1.46 <u>Event of Default</u>: shall have the meaning set forth in Section 12.1.
- 1.47 <u>Fiscal Quarter</u>: means each three month period during Borrower's Fiscal Year commencing with the first day of such Fiscal Year.
 - 1.48 Fiscal Year: shall have the meaning set forth in Section 7.15.

- 1.49 <u>GAAP</u>: generally accepted accounting principles in the United States of America, applied consistently, as in effect from time to time.
- 1.50 <u>Hazardous Substances</u>: dangerous, toxic or hazardous pollutants, contaminants, chemicals, wastes, materials or substances, as defined in or governed by the provisions of any Environmental Laws or any other federal, state or local law, statute, code, ordinance, regulation, requirement or rule relating thereto ("**Environmental Regulations**"), and also including urea formaldehyde, polychlorinated biphenyls, asbestos, asbestos-containing materials, nuclear fuel or waste, and petroleum products, or any other waste, material, substances, pollutant or contaminant which would subject an owner of property to any damages, penalties or liabilities under any applicable Environmental Regulations.
- 1.51 <u>Indebtedness</u>: means as to any Person: (a) indebtedness or liability of such Person for borrowed money, or for the deferred purchase price of property or services (including trade obligations); (b) obligations of such Person as lessee under Capital Leases; (c) obligations of such Person arising under bankers' or trade acceptance facilities; (d) all guarantees, endorsements (other than for collection or deposit in the ordinary course of business), and other contingent obligations of such Person to purchase any of the items included in this definition, to provide funds for payment, to supply funds to invest in any other Person, or otherwise to assure a creditor of another Person against loss; (e) all obligations secured by a lien on property owned by such Person, whether or not the obligations have been assumed; and (f) all obligations of such Person under any agreement providing for an interest rate swap, cap, cap and floor, contingent participation or other hedging mechanisms with respect to interest payable on any of the items described in this definition.
 - 1.52 <u>Indemnified Parties</u>: shall have the meaning set forth in Section 11.1.
- 1.53 <u>Intangible Assets</u>: means license agreements, trademarks, trade names, patents, capitalized research and development, proprietary products (the results of past research and development treated as long term assets and excluded from inventory) and goodwill (all determined in accordance with GAAP).
 - 1.54 <u>Intellectual Property</u>: shall have the meaning set forth in Section 7.18.
- 1.55 <u>Inventory</u>: means finished goods, held for sale or to be furnished under a contract of service, but excluding raw materials, work in process, or materials used or consumed in a business.
 - 1.56 <u>LC Issuance Fee</u>: shall have the meaning set forth in Subsection 3.2.3.
 - 1.57 <u>LC Request</u>: shall have the meaning set forth in Section 3.1.

9

- 1.58 <u>Letters of Credit</u>: shall have the meaning set forth in Section 3.1.
- 1.59 <u>LIBO Effective Date</u>: shall have the meaning set forth in Subsection 4.1.2.
- 1.60 <u>LIBO Rate</u>: means the rate per annum (rounded upward, if necessary, to the nearest whole 1/8 of 1%) and determined pursuant to the following formula:

adjusted by the addition of the LIBO Rate Margin.

- 1.61 <u>LIBO Rate Loan</u>: shall have the meaning set forth in Subsection 4.1.2.
- 1.62 <u>LIBO Rate Margin</u>: 275 basis points.
- 1.63 <u>LIBO Rate Period</u>: shall have the meaning set forth in Subsection 4.1.2.
- 1.64 <u>LIBO Request</u>: shall have the meaning set forth in Subsection 4.1.2.
- 1.65 <u>LIBOR Reserve Percentage</u>.: means the reserve percentage prescribed by the Board of Governors of the Federal Reserve System (or any successor) for "Eurocurrency Liabilities" (as defined in Regulation D of the Federal Reserve Board, as amended),

adjusted by Lender for expected changes in such reserve percentage during the applicable LIBO Rate Period.

- 1.66 <u>Licensing Laws</u>: shall have the meaning set forth in Section 7.4.
- 1.67 <u>Loan Documents</u>: this Credit Agreement, the Revolving Note, the Security Documents, and other documents required to grant to Lender a perfected security interest in the Collateral.
 - 1.68 <u>Loans</u>: means, collectively, all Base Rate Loans and all LIBO Rate Loans outstanding at any time.
 - 1.69 <u>Mandatory Prepayments</u>: shall have the meaning set forth in Section 5.5.
- 1.70 <u>Material Adverse Effect</u>: means: (a) a material adverse effect on the financial condition, results of operation, business or property of Borrower; (b) a material adverse effect on the ability of Borrower to perform its obligations under this Credit Agreement and the other Loan Documents; or (c) a material

10

adverse effect upon the ability of Lender to enforce its rights and remedies under the Loan Documents.

- 1.71 <u>Material Agreements</u>: shall have the meaning set forth in Section 7.16.
- 1.72 Multiemployer Plan: means a Borrower Benefit Plan defined as such in Section 3(37) of ERISA.
- 1.73 Net Worth: means total owner's equity.
- 1.74 Officer: means (a) in the case of a corporation, any chief executive officer, chief operational officer, chief financial officer, president, vice president, secretary, treasurer, assistant vice president, assistant secretary, assistant treasurer, or other corporate officer ("Corporate Officer"); (b) in the case of a partnership, any general partner, manager, and any other individual whose duties generally correspond to those performed by any Corporate Officer; (c) in the case of a limited liability company, the manager(s), any member, and any other individual whose duties generally correspond to those performed by any Corporate Officer; and (d) in the case of any other association, any individual whose duties generally correspond to those performed by any Corporate Officer.
- 1.75 <u>Organization Documents</u>: in the case of a corporation, its articles or certificate of incorporation and bylaws; in the case of a partnership, its partnership agreement and certificate of limited partnership, if applicable; in the case of a limited liability company, its articles of organization and its operating agreement.
 - 1.76 PBGC: shall have the meaning set forth in Section 7.10.
 - 1.77 <u>Permitted Encumbrance</u>: shall have the meaning set forth in Section 7.12.
- 1.78 <u>Person</u>: any individual, corporation, limited liability company, association, partnership, trust, organization, government, governmental agency, or other entity.
- 1.79 <u>Potential Default</u>: any event, other than an event described in Section 12.1(a) hereof, which with the giving of notice or lapse of time, or both, would become an Event of Default.
- 1.80 <u>Profitable Operations</u>: means Gaiam and all of the Consolidated Subsidiaries have, in the aggregate, achieved positive net income based on the most recent 10K report filed by Borrower with the Securities and Exchange Commission, or based on such comparable information as may be required by Lender in the event Borrower is not required to file a 10K report.

- 1.81 <u>Prohibited Transaction</u>: means any transaction prohibited under Section 406 of ERISA or Section 4975 of the Code.
- 1.82 <u>Regulatory Change</u>: shall have the meaning set forth in Section 13.10.
- 1.83 <u>Reportable Event</u>: means any of the events set forth in Section 4043(b) of ERISA or in the regulations thereunder.
- 1.84 <u>Required License</u>: shall have the meaning set forth in Section 7.9.
- 1.85 <u>Revolving Advance</u>: shall have the meaning set forth in Section 2.1.
- 1.86 Revolving Commitment: shall be \$15,000,000.00.
- 1.87 Revolving Facility: means the loan facility made available to Borrower under Article 2 of this Credit Agreement.
- 1.88 <u>Revolving Loan Availability Period</u>: means the period from the Closing Date until the Banking Day immediately prior to the Revolving Maturity Date.

- 1.89 Revolving Loan: means Loans under the Revolving Facility pursuant to Article 2 hereof.
- 1.90 <u>Revolving Maturity Date</u>: means July 31, 2007.
- 1.91 Revolving Note: shall have the meaning set forth in Section 2.3.
- 1.92 <u>Rules</u>: shall have the meaning set forth in Subsection 13.4.1.
- 1.93 <u>Security Documents</u>: the security agreements, mortgages, deeds of trust, financing statements, pledge agreements, leasehold assignment and consents, assignments and/or other security documents executed by Borrower in favor of Lender to secure Borrower's performance of its obligations under the Revolving Note and other Loan Documents with a first lien (subject to Permitted Encumbrances) on all accounts, inventory, and equipment of Borrower, in form and substance acceptable to Lender, as any of such documents may be amended from time to time.
- 1.94 <u>Subsidiary</u>: means with respect to any Person: (a) any corporation in which such Person, directly or indirectly, (i) owns one hundred percent (100%) or more of the outstanding stock thereof, or (ii) has the power under ordinary circumstances to elect at least a majority of the directors thereof, or (b) any partnership, association, joint venture, limited liability company, or other unincorporated organization or entity with respect to which such Person, directly or indirectly, (i) owns more than one hundred percent (100%) of the outstanding

equity interest thereof, or (ii) has the power under ordinary circumstances to directly or indirectly control the management thereof.

- 1.95 <u>Total Liabilities</u>: means with respect to any Person the aggregate amount of such Person's items properly shown as liabilities on its balance sheet, determined in accordance with GAAP.
 - 1.96 <u>Voluntary Payments</u>: shall have the meaning set forth in Section 5.4.
 - 1.97 <u>Working Capital</u>: means Current Assets minus Current Liabilities.
 - 1.98 Wire Instructions: payments to Lender shall be by wire transfer in accordance with the following instructions:

Bank: Wells Fargo Bank N.A. ABA Routing: 121000248

Ref: Gaiam Inc. Loan #2775220168-67

Account: #0763850720

Account Name: Wires in process-GL

Principal amount \$ Interest amount \$

ARTICLE 2. REVOLVING FACILITY

- 2.1 <u>Revolving Facility Loan.</u> On the terms and conditions set forth in this Credit Agreement, and so long as no Event of Default or Potential Default has occurred and is continuing, Lender agrees to make advances under the Revolving Facility ("**Revolving Advance**") from time to time during the Revolving Loan Availability Period, subject to the following:
- 2.1.1 <u>Revolving Commitment; Borrowing Base</u>. Borrower shall not be entitled to request a Revolving Advance in an amount which, when added to the outstanding amount of all previous Revolving Advances plus all Committed Advances under the Revolving Loan plus the undrawn face amount of all outstanding Letters of Credit, would exceed the lesser of: (a) the Revolving Commitment; or (b) the Borrowing Base.
- 2.2 <u>Borrowing Notice</u>. Any Borrower shall give Lender prior notice by telephone or by written notice sent by facsimile (effective upon receipt) of each request for a Revolving Advance on or before 3:30 P.M. (Mountain time) on the day of making such Revolving Advance. Each written notice must be in substantially the form of <u>Exhibit 2.2</u> hereto ("**Borrowing Notice**") and must specify (a) the amount of such Revolving Advance (subject to the provisions of Section 4.1 hereof), and (b) the proposed date of making such Revolving Advance. Each notice by telephone must provide to Lender all of the information required to be provided in a written Borrowing Notice. Borrower acknowledges that Lender is permitting telephonic requests for Revolving

13

Advances at Borrower's request and Lender shall have no liability in acting upon a telephonic request that it believes has been made by a properly authorized Person. Lender shall, not later than 3:30 P.M. (Mountain time), and upon fulfillment of the applicable conditions set forth in this Section and in Article 8 hereof, make such Revolving Advance available to such Borrower, in immediately available funds, and will transmit such funds to such Borrower's Borrower Account by wire transfer or otherwise. Each Borrower acknowledges that all Borrowers shall be jointly and severally liable for the repayment of each Revolving Advance, regardless of which Borrower has requested and received such Revolving Advance.

- 2.3 <u>Revolving Facility Promissory Note</u>. The obligations of Borrower, and each of them, to Lender under the Revolving Facility, including payment obligations with respect to all Revolving Advances made by Lender shall be evidenced by, and repaid with interest in accordance with, a single promissory note of Borrower in substantially the form of <u>Exhibit 2.3</u> hereto duly completed, in the stated maximum principal amount equal to the Revolving Commitment, payable to Lender for the account of its Applicable Lending Office, and maturing as to principal on the Revolving Maturity Date ("**Revolving Note**").
- 2.4 <u>Lender Records</u>. Lender shall record on its books and records the amount of each Revolving Advance, all payments of principal and interest under the Revolving Note, and the principal balance from time to time outstanding. Lender's record thereof shall be prima facie evidence as to all such amounts and shall be binding on Borrower absent manifest error. Notwithstanding the foregoing, Borrower will never be required to pay to Lender as principal more than the principal amount of all Revolving Advances made by Lender.
- 2.5 <u>Use of Proceeds</u>. The proceeds of the Revolving Facility will be used by Borrower for general corporate, including business acquisitions permitted hereunder, and working capital purposes and to fund draws under Letters of Credit (as provided herein), and Borrower agrees not to request or use such proceeds for any other purpose. Borrower will not, directly or indirectly, use any part of such proceeds for the purpose of purchasing or carrying any margin stock within the meaning of Regulation U of the Board of Governors or to extend credit to any Person for the purpose of purchasing or carrying any such margin stock.

ARTICLE 3. LETTER OF CREDIT FACILITY

3.1 <u>Letter of Credit Request.</u> On the terms and conditions set forth in this Credit Agreement, and so long as no Event of Default or Potential Default has occurred and is continuing (or if an Event of Default or Potential Default has occurred and is continuing, it has been waived in writing by Lender), Borrower, and each of them, may request the issuance of one or more documentary letters of credit or standby letters of credit for the benefit of Borrower ("Letters of Credit") by sending, not later than 12:00 noon (Mountain time) on a Banking

14

Day at least one (1) Banking Day before the issuance date, a request therefore via TSOL, Trade Services On Line ("LC Request") to Lender. The LC Request shall set forth the following information with respect to the requested Letter of Credit: (a) the face amount and expiry date, (b) the name of the beneficiary, (c) the terms thereof, and (d) such other information as the Lender shall request. In no event may the expiry date of a Letter of Credit be later than the Revolving Maturity Date unless at the time of its issuance (x) Lender determines that such extended expiry date is consistent with Borrower's usual and customary practices for Letters of Credit transactions during the immediately preceding twelve (12) months; and (y) Borrower agrees that, (i) this Credit Agreement and the other Loan Documents shall remain in effective and enforceable solely to enable Lender to fund any draws under each such Letter of Credit, and (ii) at Lender's option, on the Revolving Maturity Date, Borrower shall take such action as Lender shall require to grant to Lender a security interest in property approved by Lender and sufficient in liquidation value to cover the undrawn face amount of all such Letters of Credit.

- 3.2 <u>Letters of Credit.</u> No later than 1:00 P.M. (Mountain time) on the second (2nd) Banking Day after the receipt by the Lender of an LC Request (or such later date as may be specified in such LC Request), it shall issue the requested Letter of Credit for any expiry period from seven (7) days to the earlier of (a) 365 days after the date of issuance; or (b) a date which is up to 180 days after the Revolving Maturity Date, subject to the following:
- 3.2.1 <u>Limitation on Amount</u>. The face amount of the requested Letter of Credit may not, when added to the undrawn face amount of all Letters of Credit then outstanding, would exceed the lesser of (a) the Aggregate LC Commitment; or (b) an amount which, when added to the outstanding amount of all previous Revolving Advances plus all Committed Advances under the Revolving Loan, would exceed the Revolving Commitment.
- 3.2.2 <u>Availability</u>. Letters of Credit may be requested for issuance only during the period commencing on the Closing Date and ending on the Banking Day immediately preceding the Revolving Maturity Date.
- 3.2.3 <u>Fees</u>. Borrower shall at the time of issuance, renewal, or reissuance of each Letter of Credit pay to the Lender, a fee ("**LC Issuance Fee**") in the amount determined by Lender in accordance with Lender's policies and fee schedule then in effect regarding Letters of Credit and taking into account Lender's duties with respect to draws under such Letter of Credit.
- 3.2.4 <u>Treatment of Draws</u>. Each draw under a Letter of Credit shall be funded by Lender making a debit to any Borrower Account, and to the extent that there are not sufficient collected funds in any Borrower Account, by Lender making a Revolving Advance under the Revolving Facility. Each Borrower acknowledges that all Borrowers shall be jointly and severally liable for the

15

repayment of each Revolving Advance to fund a draw on a Letter of Credit, regardless of which Borrower has requested and received such Letter of Credit.

3.3 <u>Reimbursement Obligation Unconditional.</u> All draws under the Letters of Credit are absolutely, unconditionally, and irrevocably reimbursable by Borrower, and each of them individually regardless of which Borrower requested and received such Letter of Credit, and may be funded as a debit to any Borrower Account or as Revolving Advances in accordance with Subsection 3.2.4 hereof,

notwithstanding:

- (a) Any lack of validity or enforceability of the Letter of Credit, any of the documents referenced in the Letter of Credit, or any other agreement or instrument related to any such documents;
- (b) The existence of any claim, setoff, defense or other right which Borrower may have at any time against the beneficiary or any transferee of the Letter of Credit (or any person for whom the beneficiary or transferee may be acting);
- (c) Any statement, draft, certificate, or any other document presented under the Letter of Credit proving to be forged, fraudulent, invalid or insufficient in any respect, or any statement therein being untrue or inaccurate in any respect whatsoever or the draw certificate was otherwise unauthorized, it being expressly understood and agreed by Borrower that Lender shall not have any liability on account of any lack of authorization or forgery and any recovery from third parties on account of such lack of authorization or such forgery shall be the sole responsibility of Borrower; or
- (d) Payment of a draw against presentation of a draft or certificate which does not comply with the terms of the Letter of Credit, unless such payment is made as a result of the gross negligence or willful misconduct of Lender as issuer of the Letter of Credit.

ARTICLE 4. INTEREST AND FEES

- 4.1 Interest. Interest on the Revolving Note and Revolving Loan shall be calculated as follows:
- 4.1.1 <u>Base Rate Option</u>. Except for those periods during which Borrower receives a LIBO Rate Loan pursuant to Subsection 4.1.2 hereof, the outstanding principal balance under the Revolving Note shall bear interest at the Base Rate and shall be deemed a "**Base Rate Loan**".
- 4.1.2 <u>LIBO Rate Option</u>. From time to time, and so long as no Event of Default has occurred and is continuing, Gaiam may request that the entire balance owing under the Revolving Note bear interest at the LIBO Rate ("**LIBO Rate Loan**") by making a written request therefore ("**LIBO Request**") to Lender by facsimile, specifying the Banking Day, which must be at least two

16

(2) Banking Days, and not more than five (5) Banking Days, after the date of the LIBO Request, on which such balance is to begin accruing interest at the LIBO Rate ("LIBO Effective Date"). Gaiam's election to convert the balance owing under the Revolving Note to a LIBO Rate Loan shall be in effect and irrevocable for a period of thirty (30) days (or the next succeeding Banking Day in the event that such thirtieth (30th) day is not a Banking Day) commencing on, and including, the LIBO Effective Date ("LIBO Rate Period") and each LIBO Rate Period shall be automatically extended for an additional thirty (30) days unless, at least three (3) Banking Days before the end of the then current LIBO Rate Period, Gaiam makes a written request ("Conversion Request") to Lender by facsimile specifying that the entire balance owing under the Revolving Note bear interest at the Base Rate commencing on the Banking Day immediately following the end of the then current LIBO Rate Period and continuing until such time, if ever, that Gaiam makes another LIBO Request to Lender. Lender shall incur no liability in acting upon a request which it believes in good faith had been made by a properly authorized employee or other representative of Gaiam. An election for a LIBO Rate Loan shall remain in effect for the LIBO Rate Period then in effect and shall be subject to the automatic renewal provisions contained above in this Section notwithstanding the fact that the balance owing under the Revolving Note is paid down in whole or in part.

4.2 Additional Provisions for LIBO Rate Loans.

- 4.2.1 <u>Inapplicability or Unavailability of LIBO Rate</u>. If Lender at any time shall determine (which determination shall be conclusive) that for any reason adequate and reasonable means do not exist for ascertaining the LIBO Rate, or that the relevant rates of interest referred to in the definition of LIBO Rate upon the basis of which the rate of interest for LIBO Rate Loans for the LIBO Rate Period is to be determined do not adequately cover the cost to Lender of making or maintaining such LIBO Rate Loans for the LIBO Rate Period, then Lender shall promptly give notice thereof to Gaiam. If such notice is given, and until such notice has been withdrawn by Lender, then any portion of the outstanding principal balance which bears interest determined in relation to the LIBO Rate shall, subsequent to the end of the LIBO Rate Period applicable thereto, bear interest at the Base Rate, and Lender shall have no obligation to convert a Base Rate Loan to a LIBO Rate Loan.
- 4.2.2 <u>Change in Law; LIBO Rate Loan Unlawful</u>. If any law, treaty, rule, regulation or determination of a court or governmental authority or any change therein or in the interpretation or application thereof (each, a "**Change in Law**") shall make it unlawful for Lender to (a) convert a Base Rate Loan to a LIBO Rate Loan or (b) maintain all or any portion of the LIBO Rate Loans, Lender shall promptly notify Gaiam thereof in writing. In the former event, any obligation of Lender to continue a LIBO Rate Loan beyond the then effective LIBO Rate Period or to convert a Base Rate Loan to a LIBO Rate Loan shall immediately be canceled (and, in lieu thereof, shall be continued as a Base Rate Loan), and in the latter event, any such unlawful LIBO Rate Loan shall be

- 4.3 <u>Default Interest Rate</u>. All past due payments on the Revolving Note or any other Bank Debt (whether as a result of nonpayment by Borrower when due, at maturity, or upon acceleration) shall bear interest at the Default Interest Rate from and after the due date for the payment, or on the date of maturity or acceleration, as the case may be.
- 4.4 <u>Interest Calculation.</u> Interest on Base Rate Loans and on LIBO Rate Loans shall be calculated on the actual number of days the principal owing thereunder is outstanding with the daily rate calculated on the basis of a year consisting of 360 days. In calculating interest, the Advance Date shall be included and the date each payment is received shall be excluded.

ARTICLE 5. PAYMENTS

- 5.1 <u>Principal Payments</u>. Principal shall be payable under the Revolving Note on the Revolving Maturity Date, and as Voluntary Payments as provided in Section 5.4 hereof, and as Mandatory Prepayments as provided in Section 5.5 hereof.
- 5.2 <u>Interest Payments</u>. Interest on Base Rate Loans and on LIBO Rate Loans shall be payable monthly in arrears on the last Banking Day of the month and, in addition to the foregoing, interest on all Loans then accrued and unpaid shall be payable on the Revolving Maturity Date.
- 5.3 Manner of Payment. All payments, including prepayments, that Borrower is required or permitted to make under the terms of this Credit Agreement shall be made to Lender (a) in immediately available federal funds, to be received no later than 2:00 P.M. Mountain time of the Banking Day on which such payment is due by (i) wire transfer through Federal Reserve Bank, Kansas City, to the Lender in accordance with the Wire Instructions (or to such other account or by such other means as Lender may designate by notice to Borrower), (ii) Internet communication and transfer through such programs as Lender may make available from time to time, and (iii) in person or any other method as Lender makes available from time to time; and (b) without setoff or counterclaim and free and clear of and without deduction for any taxes, levies, impost, duties, charges, fees, deductions, withholding, compulsory loans, restrictions or conditions of any nature now or hereafter imposed or levied by any jurisdiction or any political subdivision thereof or taxing or other authority therein unless Borrower is required by law to make such deduction or withholding. In the event any payment whatsoever owing from Borrower to Lender under the terms of this Credit Agreement is not made when due, Lender may, at its sole option and discretion, debit any Borrower Account for the amount of such payment.

18

- 5.4 <u>Voluntary Payments</u>; Application of Voluntary Payments. Borrower shall have the right to pay ("Voluntary Payments") all or any part of the outstanding principal balance under the Loans on any Banking Day. Voluntary Payments on the Revolving Loan may be reborrowed during the Revolving Loan Availability Period under the terms and conditions and subject to the limits set forth in this Credit Agreement. Notwithstanding any of the foregoing provisions, upon the occurrence and during the continuance of an Event of Default or Potential Default, all Voluntary Payments shall be applied, as Lender in its sole discretion shall determine, to fees, costs, interest or principal indebtedness under the Revolving Note, and/or to any other Bank Debt.
- 5.5 <u>Mandatory Prepayments</u>. Borrower shall be required to make prepayments ("Mandatory Prepayments") in each of the following events: (a) in the event any of the Eligible Inventory is the subject of a Casualty Event, Borrower shall, within three (3) Banking Days of the occurrence of the Casualty Event, provide Lender with an updated Borrowing Base Certificate, taking into account the effects of the Casualty Event on the amount of Eligible Inventory and, based on such updated Borrowing Base Certificate (and upon its approval by Lender), Borrower shall make a Mandatory Prepayment as required by clause (b) of this Section; and (b) in the event that at any time the outstanding amount of all Revolving Advances, plus the amount of all Committed Advances, plus the undrawn face amount of all outstanding Letters of Credit exceeds the lesser of (i) the Revolving Commitment or (ii) the Borrowing Base, a Mandatory Prepayment in the amount of such excess. Mandatory Prepayments under clauses (a) and (b) shall be due no later than three (3) Banking Days after the occurrence of the condition requiring such Mandatory Prepayment.
- 5.6 <u>Payments on Banking Days</u>. Whenever any payment to be made hereunder shall be stated to be due on a day that is not a Banking Day, such payment shall be made on the next succeeding Banking Day and such extension of time shall be included in the computation of the interest or of any fees due hereunder, as the case may be.

ARTICLE 6. SECURITY

6.1 <u>Borrower's Assets.</u> As security for the payment and performance of all obligations of Borrower to Lender including but not limited to all principal and interest under the Revolving Note, fees, reimbursements, and all other Bank Debt or obligations under any of the Loan Documents, Borrower, and each of them individually, shall grant to, and maintain for, Lender a first lien and security interest, subject only to Permitted Encumbrances, in all of its inventory, accounts, and equipment, tangible and intangible, whether now owned or hereafter acquired, ("Collateral"), pursuant to the Security Documents. Borrower, and each of them individually, shall execute and deliver to Lender the Security Documents to evidence the security interest of Lender in the Collateral, together with such financing statements or other documents as Lender shall request. Borrower, and each of them individually, shall also execute such further

Collateral showing Lender's security interest therein; and (b) Borrower, and each of them individually, shall deliver to Lender possession of any chattel paper, instruments and securities included in the Collateral (duly endorsed to Lender's reasonable satisfaction).

ARTICLE 7. REPRESENTATIONS AND WARRANTIES

To induce Lender to make Revolving Advances and to issue Letters of Credit, and recognizing that Lender is relying thereon, Borrower, and each of them individually, represents and warrants as follows:

- 7.1 <u>Organization, Good Standing, Etc.</u> Borrower, and each of them individually: (a) is duly organized, validly existing, and in good standing under the laws of its state of formation; (b) is duly qualified to do business and is in good standing in each jurisdiction in which the transaction of its business makes such qualification necessary; and (c) has all requisite corporate and legal power to own and operate its assets and to carry on its business.
- 7.2 <u>Corporate Authority, Due Authorization; Consents.</u> Borrower, and each of them individually, has full power and authority to execute, deliver and perform under this Credit Agreement, the Revolving Note, all other Loan Documents and all other documents and agreements as contemplated by this Credit Agreement, all of which have been duly authorized. All consents or approvals of any Person that are necessary for, or are required as a condition of the execution, delivery and performance of the Loan Documents, have been obtained.
- 7.3 <u>Litigation</u>. There are, no pending legal or governmental actions, proceedings or investigations to which Borrower is a party or to which any property of Borrower is subject which could reasonably be expected to result in any Material Adverse Effect and, to Borrower's knowledge, no such actions or proceedings are threatened or contemplated by any federal, state, county, or city (or similar unit) governmental agency or any other Person.
- 7.4 No Violations. The execution, delivery and performance of the Loan Documents will not: (a) violate any provision of Borrower's Organization Documents, or any law, rule, regulation, judgment, order or ruling of any court or governmental agency; (b) violate, conflict with, result in a breach of, constitute a default under, or with the giving of notice or the expiration of time or both, constitute a default under, any existing real estate mortgage, indenture, lease, security agreement, contract, note, instrument or any other agreements or documents binding on Borrower or affecting its property; or (c) violate, conflict

20

with, result in a breach of, constitute a default under, or result in the loss of, or restriction of rights under, any Required License or any order, law, rule, or regulation under or pursuant to which any Required License was issued or is maintained ("Licensing Laws").

- 7.5 <u>Binding Agreement.</u> Each of the Loan Documents to which Borrower is a party constitutes the legal, valid and binding obligation of Borrower, enforceable in accordance with its terms, subject only to limitations on enforceability imposed by applicable bankruptcy, insolvency, reorganization, moratorium, or similar laws affecting creditors' rights generally and by general principles of equity.
- 7.6 <u>Compliance with Laws</u>. Borrower is in compliance with all federal, state, and local laws, rules, regulations, ordinances, codes and orders, including without limitation all Environmental Laws and all Licensing Laws, with respect to which noncompliance would result in a Material Adverse Effect.
- 7.7 <u>Principal Place of Business; Business Form; and Place of Formation</u>. The place of business of each Borrower, or chief executive office if it has more than one place of business, is located at the address shown on <u>Exhibit 7.7</u> hereto. The business form and state of formation of each Borrower are as shown on <u>Exhibit 7.7</u> hereto.
- 7.8 Payment of Taxes. Borrower has filed all required federal, state and local tax returns and has paid all taxes as shown to be payable by Borrower on such returns as they have become due. Borrower has paid when due all other taxes, assessments or impositions levied or assessed against Borrower or its business or properties with respect to which nonpayment could reasonably be expected to result in a Material Adverse Effect.
- 7.9 <u>Licenses and Approvals.</u> Borrower has ownership of, or license to use, or has been issued, all franchises, certificates, approvals, permits, authorities, agreements, and licenses which are used or necessary to permit it to own its properties and to conduct the business as presently being conducted except where the failure to do so could not reasonably be expected to have a Material Adverse Effect (each a "**Required License**"). Each Required License is in full force and effect, and there is no outstanding notice of cancellation or termination or, to Borrower's knowledge, any threatened cancellation or termination in connection therewith, nor has an event occurred with respect to any Required License which, with the giving of notice or passage of time or both, could result in the revocation or termination thereof or otherwise in any impairment of Borrower's rights with respect thereto, which impairment could reasonably be expected to have a Material Adverse Effect. No consent, permission, authorization, order, or license of any governmental authority, is necessary in connection with the execution, delivery, performance, or enforcement of and under the Loan Documents to which Borrower is a party,

- 7.10 Employee Benefit Plans. Exhibit 7.10 hereto sets forth as of the Closing Date a true and complete list of each Borrower Benefit Plan, broken down by the categories set forth in clauses (a) through (h) of the definition of "Borrower Benefit Plan" contained in Article 1 hereof, and each "cafeteria plan" as defined in Code Section 125, that is maintained by Borrower or any of its ERISA Affiliates or in which Borrower or any of its Subsidiaries participates or to which Borrower or any of its ERISA Affiliates is obligated to contribute, in each case as of the Closing Date. Borrower and its ERISA Affiliates are in compliance in all material respects with the Employee Retirement Income Security Act of 1974, as amended, and the regulations thereunder ("ERISA"), to the extent applicable to them, and have not received any notice to the contrary from the Pension Benefit Guaranty Corporation ("PBGC").
- 7.11 <u>Equity Investments</u>. Borrower does not now own any stock or other voting or equity interest, directly or indirectly, in any Person other than as set forth on <u>Exhibit 7.11</u>.
- 7.12 Real Property; Title to Real and Personal Property. Borrower: (a) has all real property interests, including without limitation, fee interests, leasehold interests, easements, licenses and rights of way which are necessary for the conduct of Borrower's business; (b) holds good and marketable title to all of Borrower's real property (other than rights of way, easements and similar interests in real property which in the aggregate are not material); (c) owns all of its personal property, and holds all of its leases, free and clear of any lien, pledge, restriction, or encumbrance, except as specifically identified in Exhibit 7.12 attached hereto or as permitted by Section 10.2 hereof ("Permitted Encumbrances"). All of Borrower's leases which constitute Material Agreements are in full force and effect and, where Borrower is the lessee thereunder, afford Borrower peaceful and undisturbed possession of the subject matter thereof.
- 7.13 <u>Personal Property</u>. Borrower owns or leases all tangible personal property necessary for the conduct of Borrower's business as it is being conducted; all such property is in good operating condition and repair, reasonable wear and tear excepted, and suitable in all material respects for the uses for which it is being utilized.
- 7.14 <u>Environmental Compliance</u>. Without limiting the provisions of Section 7.6 above, all property owned or leased by Borrower and all operations conducted by it are in compliance in all material respects with all Environmental Laws and Environmental Regulations, with respect to which the failure to comply would have a Material Adverse Effect.

- 7.15 <u>Fiscal Year</u>. Each fiscal year of Borrower begins on January 1 of each calendar year and ends on December 31 of each calendar year ("**Fiscal Year**").
- 7.16 <u>Material Agreements</u>. <u>Exhibit 7.16</u> hereto sets forth all material agreements of Borrower required to be filed as an exhibit to Borrower's filings with the Securities and Exchange Commission pursuant to Item 601(b)(10) of Regulation S-K of the Securities and Exchange Commission ("**Material Agreements**"). Neither Borrower nor, to Borrower's knowledge, any other party to any Material Agreement, is in default thereunder, and no facts exist which with the giving of notice or the passage of time, or both, would constitute such a default or could reasonably be expected to result in a Material Adverse Effect.
- 7.17 Regulations G, U and X. No portion of any Revolving Advance will be used for the purpose of purchasing, carrying, or making loans to finance the purchase of, any "margin security" or "margin stock" as such terms are used in Regulations G, U or X of the Board of Governors of the Federal Reserve System, 12 C.F.R. Parts 207, 221, and 224.
- 7.18 Intellectual Property. Borrower owns or licenses all registered patents, trademarks, trade names, service marks and copyrights (collectively, "Intellectual Property") that it utilizes in its business as presently being conducted, except where the failure to do so could not reasonably be expected to result in a Material Adverse Effect. Borrower is not a licensee under any written license for any patent, trademark, tradename, service mark or copyright other than shrinkwrap licenses for "off-the-shelf" software used by Borrower in the conduct of its business. The Intellectual Property registrations are in full force and effect, and Borrower has taken or caused to be taken all action, necessary to maintain the Intellectual Property registrations in full force and effect and has not taken or failed to take or cause to be taken any action which, with the giving of notice, or the expiration of time, or both, could result in any such Intellectual Property registrations being revoked, invalidated, modified, or limited, except where such revocation, invalidation, modification, or limitation could not reasonably be expected to result in a Material Adverse Effect.
- 7.19 <u>Labor Matters; Labor Agreements</u>. Except as set forth in <u>Exhibit 7.19</u> hereof: (a) There are no collective bargaining agreements or other labor agreements covering any employees of Borrower, the termination, cessation, or breach of which could reasonably be expected to result in a Material Adverse Effect, and a true and correct copy of each such agreement will be furnished to Lender upon its written request from time to time. (b) There is no organizing activity involving Borrower or pending or, to Borrower's knowledge, threatened by any labor union or group of employees. (c) There are, to Borrower's knowledge, no representation proceedings pending or threatened with the National Labor Relations Board, and no labor organization or group of employees of Borrower, has made a pending demand for recognition. (d) There are no complaints or charges against Borrower or pending or, to Borrower's

23

knowledge threatened to be filed with any federal, state, local or foreign court, governmental agency or arbitrator based on, arising out of, in connection with, or otherwise relating to the employment or termination of employment by Borrower, of any individual. (e) There are no strikes or other labor disputes against Borrower that are pending or, to Borrower's knowledge, threatened. (f) Hours worked by and payment made to employees of Borrower have not been in violation of the Fair Labor Standards Act (29 U.S.C. § 201 et seq.) or any other applicable law dealing with such matters. The representations made in clauses (b) through (f) of this Section are made with respect to those occurrences described which could, considered in the aggregate, reasonably be expected to have a Material Adverse Effect.

- 7.20 <u>Borrowing Base Certificate; Compliance Certificate</u>. The Borrowing Base Certificate and the Compliance Certificate are accurate as of the date indicated therein.
- 7.21 <u>Disclosure</u>. The representations and warranties contained in this Article 7 and in the other Loan Documents or in any financial statements or projections provided to Lender do not contain any untrue statement of a material fact or omit to state a material fact necessary to make such representations or warranties not misleading.

ARTICLE 8. CONDITIONS TO REVOLVING ADVANCES

- 8.1 <u>Conditions to Closing.</u> The obligation of Lender to make any Revolving Advances and to issue Letters of Credit hereunder is subject to satisfaction, in the sole discretion of Lender, of each of the following conditions precedent prior to the first Revolving Advance or the first issuance of a Letter of Credit (except as specifically provided otherwise in Subsection 8.1.2):
 - 8.1.1 Loan Documents. Lender shall have received duly executed originals of the Loan Documents.
- 8.1.2 <u>Searches; UCC Filings; Recordings</u>. Lender shall have received current searches of appropriate filing offices (dated no more than thirty (30) days prior to the Closing Date) showing that (a) no state or federal tax liens have been filed which remain in effect against Borrower, or any of them, or any of the Collateral; (b) except with respect to Permitted Encumbrances no financing statements have been filed by any Person except to perfect the security interests required by this Credit Agreement, which remain in effect against Borrower, or any of them, or any of the Collateral or other assets of any Borrower; (c) all financing statements necessary to perfect the security interests in the Collateral granted to Lender under the Loan Documents have been filed or recorded, to the extent such security interests are capable of being perfected by such filing and/or recording; and (d) all of the Loan Documents required to be recorded or filed to perfect the security interests and liens granted therein in the Collateral shall be so recorded and filed.

24

- 8.1.3 <u>Approvals</u>. Lender shall have received evidence satisfactory to it that all consents and approvals of governmental authorities and third parties which are with respect to Borrower, necessary for, or required as a condition of: (a) the validity and enforceability of the Loan Documents to which it is a party; and (b) creation and perfection of, and realization on, Lender's lien on the Collateral, have, in each such case, been obtained and are in full force and effect.
- 8.1.4 <u>Organizational Documents</u>. Lender shall have received for each Borrower: (a) good standing certificates, dated no more than thirty (30) days prior to the Closing Date, for such Borrower for its state of formation (to the extent such certification is available) and for each state where its operations require qualification or authorization to transact business; (b) a copy of the Organization Documents of such Borrower certified to by the appropriate governmental official in its state of formation with which they must be filed (to the extent such certification is available); and (c) a copy of the Organization Documents of such Borrower, certified as true and complete by the Secretary or Assistant Secretary, Managing Partner, or Manager, of such Borrower, as applicable, to the extent certification by the appropriate governmental official in its state of formation with which they must be filed is not available.
- 8.1.5 <u>Evidence of Authorization</u>. Lender shall have received in form and substance satisfactory to Lender documents evidencing all corporate or other authorizing action taken by each Borrower to authorize (including the specific names and titles of the persons authorized to so act (each an "Authorized Officer")) the execution, delivery and performance of the Loan Documents to which it is a party, certified to be true and correct by the Secretary or Assistant Secretary, Managing Partner, or Manager, of such Borrower, as applicable.
- 8.1.6 <u>Legal Opinion for Borrower</u>. Lender shall have received opinions of counsel for each Borrower (who shall be reasonably acceptable to Lender), in form and content acceptable to Lender and addressed to Lender and its present and future assigns and participants.
- 8.1.7 <u>Evidence of Insurance</u>. Borrower shall have provided Lender with insurance certificates and such other evidence, in form and substance satisfactory to Lender, of all insurance required to be maintained by it under the Loan Documents.
- 8.1.8 <u>Material Agreements</u>. Lender shall have received copies of those Material Agreements as Lender may request in its sole discretion.
- 8.1.9 <u>Borrowing Base Certificate</u>. Borrower shall have provided to Lender a proper Borrowing Base Certificate accurate as of June 30, 2005.
- 8.1.10 <u>Inventory Report</u>. Borrower shall have provided to Lender a proper Inventory Report accurate as of June 30, 2005.

25

8.1.11 [Intentionally Omitted]

8.1.12 <u>No Material Change</u>. No change shall have occurred in the condition or operations of Borrower since June 30, 2005 which could reasonably be expected to result in a Material Adverse Effect.

- 8.1.13 <u>Fees and Expenses</u>. Borrower shall have paid Lender, by wire transfer of immediately available federal funds all fees owing to Lender that are due on the Closing Date, and all expenses owing pursuant to Section 13.1 hereof.
- 8.1.14 <u>Accounts Aging Report</u> . Borrower shall have provided to Lender a proper Accounts Aging Report which satisfies the requirements set forth in Subsection 9.2.15 hereof and which is accurate as of June 30, 2005.
- 8.1.15 Operating Accounts; Treasury Management Services. Borrower, and each of them, shall have taken such action as Lender shall reasonably require to insure that all of Borrower's material operating accounts and related treasury management services are maintained with, and/or being handled by, Lender.
- 8.1.16 <u>Further Assurances</u>. Borrower shall have provided and/or executed and delivered to Lender such further assignments, documents or financing statements, in form and substance satisfactory to Lender, that Borrower is to execute and/or deliver pursuant to the terms of the Loan Documents or as Lender may reasonably request.
- 8.2 <u>Conditions to Revolving Advance and Issuance of Letters of Credit</u>. Lender's obligation to fund each Revolving Advance and to issue Letters of Credit is subject to the satisfaction, in the sole discretion of Lender, of each of the following conditions precedent, as well as those set forth in Section 8.1 hereof, and each Borrowing Notice or LC Request submitted by Borrower shall constitute a representation by Borrower, upon which Lender may rely, that the conditions set forth in Subsections 8.2.3, 8.2.4, and 8.2.5 hereof have been satisfied:
- 8.2.1 <u>Borrowing Notice; LC Request</u>. Borrower shall have provided to Lender a Borrowing Notice that satisfies the requirements of Section 2.2 hereof or an LC Request that satisfies the requirements of Section 3.1 hereof, as applicable.
- 8.2.2 <u>Possession of Documents</u>. Lender shall have received possession of all instruments or securities, if any, with respect to which possession is required to perfect Lender' security interest therein.
- 8.2.3 <u>Default</u>. As of the Advance Date no Event of Default or Potential Default shall have occurred and be continuing, and the disbursing of

the amount of the Revolving Advance requested in the Borrowing Notice shall not result in an Event of Default or Potential Default.

- 8.2.4 <u>Representations and Warranties</u>. The representations and warranties of Borrower contained in each of the Loan Documents to which it is a party and all exhibits thereto, subject to updates provided from time to time by Borrower and (except for updates of disclosure items listed on Exhibits 7.11 and 7.16 to reflect investments and other events and transactions permitted by this Agreement) approved in writing by Lender in its reasonable discretion, shall be true and correct in all material respects on and as of the date on which the Revolving Advance is to be made as though made on such date. Borrower shall have paid Lender, by wire transfer of immediately available U.S. funds all fees which are then due and payable, including all expenses owing pursuant to Section 13.1 hereof.
- 8.2.5 <u>Availability Period</u>. The Borrowing Notice does not specify an Advance Date which is later than the Banking Day immediately prior to the last day of the Revolving Loan Availability Period.
- 8.2.6 <u>No Material Change</u>. No change shall have occurred in the condition or operations of Borrower since the date of the financial statements most recently provided by Borrower to Lender pursuant to Subsection 9.2.1 or 9.2.2 hereof, as applicable, which, when considered in the aggregate, could reasonably be expected to result in a Material Adverse Effect.

ARTICLE 9. AFFIRMATIVE COVENANTS

From and after the date of this Credit Agreement and until the Bank Debt is indefeasibly paid in full, all Letters of Credit have expired, and Lender has no obligation to make any Revolving Advance and no obligation to issue Letters of Credit hereunder, Borrower, and each of them, agrees that it will observe and comply with the following covenants for the benefit of Lender:

- 9.1 <u>Books and Records.</u> Borrower shall at all times keep proper books of record and account, in which correct and complete entries shall be made of all its dealings, in accordance with GAAP.
 - 9.2 <u>Reports and Notices</u>. Borrower shall provide to Lender the following reports, information and notices:
- 9.2.1 <u>Annual Financial Statements</u>. As soon as available, but in no event later than one hundred and fifty (150) days after the end of any Fiscal Year of Borrower occurring during the term hereof annual financial statements of Gaiam and each of the Consolidated Subsidiaries, prepared in accordance with GAAP consistently applied which shall: (a) be audited by independent certified public accountants selected by Borrower which are reasonably acceptable to Lender; (b) be accompanied by a report of such accountants containing an opinion reasonably acceptable to Lender; and (c) include a balance sheet, an

prepared in reasonable detail and in comparative form; and (e) be accompanied by a Compliance Certificate. The requirements of clauses (a) through (d) of this Subsection will be satisfied if Borrower provides Lender with the financial statement Borrower files with the Securities and Exchange Commission with its 10K report, so long as Borrower is required to makes such a filing and the filing is made on a timely basis.

- 9.2.2 Quarterly Financial Statements. As soon as available but in no event more than ninety (90) days after the end of each Fiscal Quarter the following financial statements or other information concerning the operations of Gaiam and each of the Consolidated Subsidiaries, prepared in accordance with GAAP consistently applied: (a) a balance sheet; (b) an income statement; (c) a statement of cash flows; (d) a statement of stockholders' equity, for such Fiscal Quarter and for the Fiscal Year to date; and (e) such other quarterly statements as Lender may reasonably request, which quarterly statements requested under this clause (e) shall include any and all notes and schedules thereto. Such quarterly financial statements required pursuant to this Subsection shall be accompanied by a Compliance Certificate. The requirements of clauses (a) through (d) of this Subsection will be satisfied if Borrower provides Lender with the financial statement Borrower files with the Securities and Exchange Commission with its 10Q report, so long as Borrower is required to make such a filing and the filing is made no more than forty-five (45) days after the end of each Fiscal Quarter.
- 9.2.3 Notice of Default. As soon as the existence of any Event of Default or Potential Default becomes known to any Officer of Borrower, Borrower shall promptly give Lender written notice of such Event of Default or Potential Default, the nature and status thereof, and the action being taken or proposed to be taken with respect thereto.
- 9.2.4 <u>Notice of Certain Changes</u>. Borrower shall notify Lender at least ten (10) Banking Days prior to the occurrence of any change in the name, business form, or place of formation, of Borrower in order to maintain the perfected status of Lender's first lien and security interest (subject only to Permitted Encumbrances) in the Collateral.
- 9.2.5 <u>Notice of Litigation</u>. Borrower shall promptly notify Lender in writing of all litigation in which Borrower is a party, and which either: (a) involves an amount of \$200,000 or more, singly or in the aggregate at any time; or (b) could reasonably be expected to result in a Material Adverse Effect with respect to Borrower.
- 9.2.6 <u>Notice of Material Adverse Effect</u>. Promptly after any Officer of Borrower obtains knowledge thereof, notice of any matter which has resulted or could reasonably be expected to result in a Material Adverse Effect.

- 9.2.7 <u>Notice of Environmental Litigation</u>. Without limiting the provisions of Subsection 9.2.5 hereof, promptly after Borrower's receipt thereof, notice of the receipt of all pleadings, orders, complaints, indictments, or other communication alleging a condition that may require Borrower to undertake or to contribute to a cleanup or other response under Environmental Regulations, or which seeks penalties, damages, injunctive relief, or criminal sanctions related to alleged violations of such laws, or which claims personal injury or property damage to any person as a result of environmental factors or conditions or which, if adversely determined, could reasonably be expected to have a Material Adverse Effect.
- 9.2.8 <u>Regulatory and Other Notices</u>. Promptly after Borrower's receipt thereof, copies of any notices or other communications received from any governmental authority with respect to any matter or proceeding the effect of which could reasonably be expected to have a Material Adverse Effect.
- 9.2.9 <u>Adverse Action Regarding Required Licenses</u>. In the event Borrower learns that any petition, action, investigation, notice of violation or apparent liability, notice of forfeiture, order to show cause, complaint or proceeding is pending, or, to the best of the knowledge of any officer of Borrower, threatened, to seek to revoke, cancel, suspend, modify, or limit any of the Required Licenses, Borrower shall provide Lender with prompt written notice thereof and shall take, or cause to be taken, all reasonable measures to contest such action in good faith.
- 9.2.10 ERISA Reports. As soon as possible and in any event within twenty (20) days after Borrower or any ERISA Affiliate knows or has reason to know that any Reportable Event or Prohibited Transaction has occurred with respect to any Borrower Benefit Plan or that the PBGC or Borrower or any ERISA Affiliate has instituted or will institute proceedings under Title IV of ERISA to terminate any Borrower Pension Plan, or that Borrower or any ERISA Affiliate has completely or partially withdrawn from a Multiemployer Plan, or that a Borrower Benefit Plan which is a Multiemployer Plan is in reorganization (within the meaning of Section 4241 of ERISA), is insolvent (within the meaning of Section 4245 of ERISA) or is terminating, a certificate of the Chief Financial Officer of Borrower or such ERISA Affiliate setting forth details as to such Reportable Event or Prohibited Transaction or Borrower Pension Plan termination or withdrawal or reorganization or insolvency and the action Borrower or such ERISA Affiliate proposes to take with respect thereto, provided, however, that notwithstanding the foregoing, no reporting is required under this subsection unless the matter(s), individually or in the aggregate, result, or could be reasonably expected to result, in aggregate obligations or liabilities of Borrower and/or the ERISA Affiliates in excess of \$100,000.
- 9.2.11 <u>Borrowing Base Certificate</u>. Borrower shall, no later than the thirtieth (30th) day of each Fiscal Quarter, provide to Lender a Borrowing Base Certificate, certified to by Gaiam's chief financial officer, effective as of

by each Borrower individually.

- 9.2.12 <u>Quarterly Inventory Report</u>. Borrower shall, no later than the thirtieth (30th) day of each Fiscal Quarter, provide to Lender an Inventory Report, certified to by Gaiam's chief financial officer, effective as of the last day of the previous Fiscal Quarter, and including a breakdown as to amounts of Inventory owned by each Borrower individually.
- 9.2.13 <u>Annual Projections</u>. Borrower shall, no later than ninety (90) days after the end of each Fiscal Year, provide to Lender projections for Borrower's forecasted operations during such Fiscal Year showing, on a month-by-month and year to date cumulative basis, including forecasted income, expenses, and borrowing needs. Such projections shall be accompanied by a certificate of Borrower's Authorized Officer certifying that, (a) at the time of submission, such projections are based upon reasonable estimates and assumptions, all of which were fair in light of then-current conditions, (b) such projections were prepared on the basis of the assumptions stated therein, and reflect the reasonable estimate of Borrower of the results of operations and other information projected therein; (c) to the knowledge of the Person providing the certification, as of the date thereof (i) there have been no changes in the estimates and assumptions used in preparing the projections which, if taken into account, would, in the aggregate, have a material adverse effect on such projections, and (ii) such estimates and assumptions are fair in light of the current conditions; and (d) no undisclosed facts exist which, if taken into account, would, in the aggregate, have a material adverse effect on the projections.
- 9.2.14 <u>SEC Filings and Reports</u>. Promptly upon their becoming available, copies of all registration statements and regular periodic reports, if any, which Borrower shall have filed with the Securities and Exchange Commission or any governmental agency substituted therefor, or any national securities exchange, including copies of Borrower's form 10-K annual report, form 10-Q quarterly report and any Form 8-K report filed with the Securities and Exchange Commission.
- 9.2.15 Accounts Aging Report. Borrower shall, no later than the thirtieth (30th) day of each Fiscal Quarter, provide to Lender a report ("Accounts Aging Report"), certified to by Gaiam's chief financial officer, effective as of the last day of the previous Fiscal Quarter, showing for each Borrower whose Accounts Receivable are included in the Borrowing Base: (a) aggregate balances of all Accounts Receivable owing on a current, 30-day, 60-day, 90-day and over 90-day basis, (b) by name of account debtor, those accounts (and amounts) which are more than 90-days past due, (c) by name of account debtor, those of Accounts Receivable owing (and amounts) as to which the account debtor has (i) threatened or asserted a set-off, defense or adverse claim

- or (ii) threatened or asserted a refusal to pay, (d) individually by name and amount owing all account debtors with respect to which twenty-five percent (25%) or more of their total outstanding balance of Accounts Receivable owing constitutes Delinquent Accounts, and (e) by name and amount owing all account debtors whose outstanding balances of Accounts Receivable owing individually constitute in excess of twenty-five percent (25%) of Borrower's total of Accounts Receivable owing at the time of the report
- 9.2.16 <u>Additional Information</u>. With reasonable promptness such additional financial information or other documentation as Lender may reasonably request.
- 9.3 <u>Maintenance of Existence and Qualification</u>. Each Borrower shall maintain its existence in good standing under the laws of its state of formation. Each Borrower shall qualify and remain qualified to do business in each jurisdiction in which such qualification is necessary or desirable in view of its business, operations and properties.
- 9.4 <u>Compliance with Legal Requirements and Agreements</u>. Borrower shall: (a) comply with all laws, rules, regulations and orders applicable to Borrower or its business; and (b) comply with all agreements, indentures, mortgages, and other instruments to which it is a party or by which it or any of its property is bound; provided, however, that the failure of Borrower to comply with this sentence in any instance not directly involving Lender shall not constitute an Event of Default unless such failure could reasonably be expected to have a Material Adverse Effect.
- 9.5 <u>Compliance with Environmental Laws.</u> Without limiting the provisions of Section 9.4 of this Credit Agreement, Borrower shall comply in all material respects with, and take all reasonable steps necessary to cause all persons occupying or present on any properties owned or leased by Borrower to comply with, all Environmental Laws and Environmental Regulations, the failure to comply with which could reasonably be expected to have a Material Adverse Effect.
- 9.6 <u>Taxes</u>. Borrower shall cause to be paid when due all taxes, assessments, and other governmental charges upon, and payable by, it, its income, its sales, its properties, and federal and state taxes withheld from its employees' earnings, unless such taxes, assessments, or other governmental charges shall be contested in good faith by appropriate actions or legal proceedings and Borrower shall establish adequate reserves therefor in accordance with GAAP.
- 9.7 <u>Insurance</u>. Borrower shall keep its assets, including the Collateral, insured at all times by an insurance carrier or carriers reasonably acceptable to Lender which have an AI rating by the current BEST Key Rating Guide, against all risks covered by a special form policy (and including flood (for property

business. Borrower shall also maintain fidelity coverage (including employee dishonesty) on such officers and employees and in such amounts as Lender shall reasonably specify, or in the absence of any such specification, as customarily carried by Persons engaged in comparable businesses and comparably situated. Such insurance policies shall contain such reasonable endorsements as Lender shall from time to time require and all liability policies shall name Lender as an additional insured as its interests may appear. All such casualty insurance policies shall be endorsed with a loss payable clause, as appropriate, in favor of Lender. Certificates of all insurance referred to in this Section satisfactory to Lender shall be delivered to and held by Lender, and the policy or policies evidencing all such insurance shall be provided to Lender upon Lender's request therefor. All such insurance policies shall contain a provision requiring at least ten (10) days' notice to Lender prior to any cancellation for non-payment of premiums and at least forty-five (45) days' notice to Lender of cancellation for any other reason or of modification or non-renewal. No later than forty (40) days prior to expiration, Borrower shall give Lender (a) satisfactory written evidence of renewal of all such policies with premiums paid; or (b) a written report as to the steps being taken by Borrower to renew or replace all such policies, provided that notwithstanding the receipt of such written report, Lender may at any time thereafter give Borrower written notice to provide Lender with such evidence as described in clause (a), in which case Borrower must do so within ten (10) days of such notice. Borrower agrees to pay all premiums on such insurance as they become due, and will not permit any condition to exist on or with respect to its assets which would wholly or partially invalidate any insurance thereon. Effective upon the occurrence of an Event of Default, all of Borrower's right, title and interest in and to all such policies and any unearned premiums paid thereon are hereby assigned to Lender who shall have the right, but not the obligation, to assign the same to any purchaser of the Collateral at any foreclosure sale. Borrower shall give immediate written notice to the insurance carrier and Lender of any loss. Borrower hereby authorizes and empowers Lender upon the occurrence and during the continuation of an Event of Default, at Lender's option and in Lender's sole discretion, to act as attorney-in-fact for Borrower to make proof of loss, to adjust and compromise any claim under insurance policies, to collect and receive insurance proceeds, and to deduct therefrom Lender's expenses incurred in the collection of such proceeds, all as it shall relate to the Collateral.

9.8 <u>Title to Assets and Maintenance</u>. Borrower shall defend and maintain title to all its properties and assets, including the Collateral. Borrower shall keep its assets, both real and personal, including the Collateral, in good order and condition consistent with industry practice and shall make all

32

necessary repairs, replacements and improvements so that its business may be properly and advantageously conducted.

- 9.9 Payment of Liabilities. Borrower shall pay all liabilities (including, without limitation: (a) any indebtedness for borrowed money or for the deferred purchase price of property or services; (b) any obligations under leases which have or should have been characterized as capitalized leases, as determined in accordance with GAAP; and (c) any contingent liabilities, such as guaranties, for the obligations of others relating to indebtedness for borrowed money or for the deferred purchase price of property or services or relating to obligations under leases which have or should have been characterized as capitalized leases, as determined in accordance with GAAP) as they become due beyond any period of grace under the instrument creating such liabilities, unless (with the exception of the Bank Debt) they are being contested in good faith by appropriate actions or legal proceedings, Borrower establishes adequate reserves therefor in accordance with GAAP, and such contesting will not result in a Material Adverse Effect.
- 9.10 <u>Further Assurances</u>. Borrower shall, as may be required from time to time by Lender, provide such documents as may be necessary or desirable in the judgment of Lender to verify the existence and perfection of the security interest in the Collateral granted to Lender.
- 9.11 <u>Inspection</u>. Borrower shall permit Lender or its agents, during normal business hours or at such other times as the parties may agree, to examine Borrower's properties (including, but without limitation, the Collateral), books, and records, and to discuss Borrower's affairs, finances, operations, and accounts with its respective officers, directors, employees, and independent certified public accountants.
- 9.12 <u>Required Licenses; Permits; Etc.</u> Borrower shall duly and lawfully obtain and maintain in full force and effect all Required Licenses as appropriate for the business being conducted and properties owned by Borrower at any given time.
- 9.13 <u>ERISA</u>. Borrower shall make or cause to be made, and cause each ERISA Affiliate to make or cause to be made, all payments or contributions to all Borrower Benefit Plans covered by Title IV of ERISA, which are necessary to enable those Borrower Benefit Plans to continuously meet all minimum funding standards or requirements.
- 9.14 <u>Financial Covenants</u>. Borrower shall maintain the following financial covenants on a consolidated basis of the operations of Gaiam and each Consolidated Subsidiary:
- 9.14.1 Net Worth. At, and measured as of, the end of each Fiscal Quarter Net Worth shall be no less than: (a) \$40,000,000 for the period from the

33

Closing Date through and including September 30, 2005; and (b) \$47,000,000 as of December 31, 2005 and thereafter.

9.14.2 <u>Current Ratio</u>. At, and measured as of, the end of each Fiscal Quarter, a ratio of Current Assets, divided by Current Liabilities of equal no less than 1.35 to 1.0.

- 9.14.3 <u>Total Liabilities to Net Worth</u>. At, and measured as of, the end of each Fiscal Quarter, a ratio of Total Liabilities, divided by Net Worth (in each case as determined in accordance with GAAP), of greater than 1.00 to 1.0.
 - 9.14.4 Minimum Net Income. At, and measured as of, the end of each Fiscal Year, Profitable Operations.
- 9.15 <u>Change of Management.</u> Jirka Rysavy shall at all times be Gaiam's Chief Executive Officer and/or Chairman (defined as the person who proposes and implements all significant policies regarding Gaiam's operations and who directs the day to day operations of Gaiam).

ARTICLE 10. NEGATIVE COVENANTS

From and after the date of this Credit Agreement and until the Bank Debt is indefeasibly paid in full, all Letters of Credit have expired, and Lender has no obligation to make Revolving Advances and no obligation to issue Letters of Credit hereunder, Borrower, and each of them, agrees to observe and comply with the following covenants for the benefit of Lender:

- 10.1 <u>Borrowing.</u> Borrower shall not create, incur, assume or permit to exist, directly or indirectly, any Indebtedness, except for: (a) Indebtedness of Borrower arising under this Credit Agreement and the other Loan Documents; (b) current operating liabilities (other than for borrowed money) incurred in the ordinary course of business; (c) Indebtedness on the date hereof as set forth in <u>Exhibit 10.1</u> attached hereto; (d) Indebtedness subordinated to all Bank Debt in a manner acceptable to Lender in its sole discretion; (e) other Indebtedness, including, without limitation, Indebtedness arising under guarantees permitted under Section 10.4 hereof and Indebtedness arising under Capital Leases, in a maximum amount of principal outstanding at any one time of \$5,000,000.00; (f) indebtedness constituting any refinancing or refunding of indebtedness described in clauses (d) and (e) of this Section, provided that the principal amount thereof does not increase as a result of any such refinancing or refunding from the balance owing on the date hereof or on the date of such refinancing or refunding, whichever is lower; and (g) investments permitted by Section 10.6 (to the extent such investments constitute Indebtedness by virtue of being an obligation to supply funds to invest in any Person).
- 10.2 <u>Liens</u>. Borrower will not create, incur, assume or suffer to exist any mortgage, pledge, lien, charge or other encumbrance on, or any security interest in, any of its real or personal properties (including, without limitation,

34

leasehold interests, leasehold improvements and any other interest in real property or fixtures), now owned or hereafter acquired, except:

- (a) Liens for taxes or other governmental charges which are not due or remain payable without penalty, or are being contested in good faith by appropriate actions or proceedings; provided that such reserves or other appropriate provisions, if any, as shall be required by GAAP, shall have been made for such taxes or other governmental charges;
- (b) Deposits or pledges to secure workmen's compensation, unemployment insurance, old age benefits or other social security obligations or in connection with or to secure the performance of bids, tenders, trade contracts or leases or to secure statutory obligations or surety or appeal bonds or other pledges or deposits of like nature and all in the ordinary course of business;
- (c) Mechanics', carriers', workmen's, repairmen's or other like liens arising in the ordinary course of business in respect of obligations not yet due or which are being contested in good faith and by appropriate proceedings;
- (d) Easements, rights-of-way, zoning restrictions and other similar matters incidental to the ownership of property which do not in the aggregate materially detract from the value of such property or assets or materially impair their use in the operation of the business of Borrower;
- (e) Purchase money security interests in property (including any security interest created in a lease transaction); provided that (i) such property is used in the ordinary course of Borrower's business, (ii) such security interests shall attach only to the property so purchased (or, if applicable, leased), (iii) the amount of the purchase money financing so secured does not exceed the amount permitted under Section 10.1 hereof, and (iv) the purchase (or, if applicable, lease) occurred subsequent to the Closing Date;
- (f) All precautionary filings of financing statements under the Uniform Commercial Code which cover property that is made available to or used by Borrower pursuant to the terms of an operating lease or Capital Lease with an aggregate asset value of \$1,500,000.00; and
 - (g) Liens in existence on the date hereof as set forth in Exhibit 10.2 attached hereto.
- 10.3 <u>Sale of Assets.</u> Borrower will not sell, convey, assign, lease or otherwise transfer or dispose of, voluntarily, by operation of law or otherwise, any of its assets to any Person, except that; (a) Borrower may sell inventory, equipment and fixtures disposed of in the ordinary course of its business; (b) Borrower may dispose of inventory and equipment which is worn out or obsolete or no longer used or useful by Borrower in its business so long as no Event of Default has occurred and is continuing.

35

10.4 <u>Liabilities of Others</u>. Borrower shall not assume, guarantee, become liable as a surety, endorse, contingently agree to purchase, or otherwise be or become liable, directly or indirectly (including, but not limited to, by means of a maintenance agreement, an

asset or stock purchase agreement, or any other agreement designed to ensure any creditor against loss), for or on account of the obligation of any Person, except (a) by the endorsement of negotiable instruments for deposit or collection or similar transactions in the ordinary course of the Borrower's business; (b) those liabilities to be assumed under the Asset Purchase Agreement between Gaiam, Inc. and GT Brands LLC, GT Merchandising & Licensing LLC, Gym Time, LLC, BSBP Productions LLC, and GoodTimes Entertainment LLC, dated July 8, 2005 ("GoodTimes Agreement"); (c) up to an aggregate of \$10,000,000.00 in connection with acquisitions permitted pursuant to Section 10.5 hereof; and (d) subject to the limitations contained in Section 10.1(f) hereof, guarantees made from time to time by Borrower in the ordinary course of its business.

- 10.5 Merger; Acquisitions; Business Form; Etc. Borrower shall not merge or consolidate with any entity, or acquire all or substantially all of the assets of any Person or entity, change its business form, or commence operations under any other name, organization, or entity, including any joint venture; provided however that Borrower may (a) make acquisitions pursuant to which payment is made solely in stock or equity interests in Borrower; (b) complete the asset purchase contemplated in the GoodTimes Agreement; (c) in addition to acquisitions permitted under clauses (a) and (b) of this Section, make acquisitions up to, for all Borrowers in the aggregate, \$10,000,000.00 in consideration paid in any Fiscal Year; (d) form new Subsidiaries; and (e) merge or consolidate one Borrower into another Borrower, provided that no later than ten (10) Business Days after the creation of a new Subsidiary or any merger or consolidation, as described in clauses (d) or (e), Borrower shall (i) provide Lender with written notification of such transactions (including a detailed description, the name of any newly formed Person, any change in the name of any existing Person, the business form of any newly formed Person or change in business form of any existing Person, and the place of location (as that term is used in the applicable Uniform Commercial Code) of any newly formed Person or any change of location of any existing Person, and (ii) take such action (or cause such newly formed Person to take such action) as Lender directs in order to create and/or perfect Lender's security interest in any Collateral.
- 10.6 <u>Investments</u>. Borrower will not own, purchase or acquire any stock, obligations or securities of, or any other interest in, or make any capital contribution to, any Person, except that Borrower may own, purchase or acquire:
- (a) Commercial paper maturing not in excess of one year from the date of acquisition and rated P1 by Moody's Investors Service, Inc. or A1 by Standard & Poor's Corporation on the date of acquisition;

36

- (b) Certificates of deposit in North American commercial banks rated C or better by Keefe, Bruyette & Woods, Inc. or 3 or better by Cates Consulting Analysts, maturing not in excess of one year from the date of acquisition;
- (c) Obligations of the United States government or any agency thereof, the obligations of which are guaranteed by the United States government, maturing, in each case, not in excess of one year from the date of acquisition; and
- (d) Repurchase agreements of any bank or trust company incorporated under the laws of the United States of America or any state thereof and fully secured by a pledge of obligations issued or fully and unconditionally guaranteed by the United States government; and
- (e) The Investments described on Exhibit 10.6 hereto and owned as of the Closing Date, investments permitted by Section 10.5 hereof, and up to an aggregate of \$5,000,000.00 in Investments acquired by Borrower subsequent to the Closing Date' provided that such additional \$5,000,000.00 in Investments shall not include Investments which would otherwise be prohibited by Section 10.5 hereof.
- 10.7 <u>Loans</u>. Borrower shall not lend or advance money, credit, or property to any Person, except for trade credit extended in the ordinary course of business and loans or advances to employees not in excess of \$50,000.00 for any such employee and an aggregate at any time of \$250,000.00 for all employees.
- 10.8 <u>Transactions With Related Parties</u>. Borrower shall not purchase, acquire, or sell any equipment, other personal property, real property or services from or to any Affiliate, except in the ordinary course and pursuant to the reasonable requirements of Borrower's business and upon fair and reasonable terms no less favorable than would be obtained by Borrower in a comparable arm's-length transaction with an unrelated Person.
- 10.9 <u>ERISA</u>. Borrower shall not: (a) adopt, maintain or become obligated to contribute to any Borrower Benefit Plan not disclosed on <u>Exhibit 7.10</u> hereto without the prior written consent of Lender, which shall not be unreasonably withheld; (b) engage in or permit any transaction which results in a Prohibited Transaction or in the imposition of an excise tax pursuant to Section 4975 of the Code; (c) engage in or permit any transaction or other event which results in a Reportable Event for any Borrower Pension Plan; (d) fail to make full payment when due of all amounts which, under the provisions of any Borrower Benefit Plan, Borrower is required to pay as contributions thereto; (e) permit to exist any "accumulated funding deficiency" (as such term is defined in Section 302 of ERISA) in excess of \$25,000.00, whether or not waived, with respect to any Borrower Pension Plan; (f) fail to make any payments to any Multiemployer Plan that Borrower may be required to make under any agreement

37

relating to such Multiemployer Plan or any law pertaining thereto; or (g) terminate any Borrower Pension Plan without the prior written consent of Lender. As used in this Section, all terms enclosed in quotation marks shall have the meanings set forth in ERISA. Borrower's failure to comply with any of the foregoing provisions of this Section shall not constitute a breach of this Credit Agreement or an Event of

Default unless such failure could reasonably be expected to result in a Material Adverse Effect.

- 10.10 <u>Payment of Dividends</u>. Borrower shall not, directly or indirectly, declare or pay any dividends on account of any shares of any class of its capital stock now or hereafter outstanding, or set aside or otherwise deposit or invest any sums for such purpose.
 - 10.11 Change in Fiscal Year. Borrower shall not change its Fiscal Year.
- 10.12 <u>Capital Expenditures</u>. Borrower shall not make any Capital Expenditures in excess of \$10,000,000.00 in the aggregate in any Fiscal Year.
- 10.13 <u>Intercompany Transactions</u>. Notwithstanding anything to the contrary contained in Article 10, so long as no Event of Default has occurred and is continuing, a Borrower may (a) create or incur Indebtedness to another Borrower, (b) sell, convey, assign, lease or otherwise transfer any of its assets to another Borrower, (c) guarantee or otherwise become liable for an obligation of another Borrower that is otherwise permitted under this Agreement, (d) make investments in, or lend or advance money, credit or property to, another Borrower, (e) enter into transactions with another Borrower without complying with the requirements set forth in Section 10.8, and (f) declare or pay dividends or otherwise make distributions to another Borrower.

ARTICLE 11. INDEMNIFICATION

11.1 General; Stamp Taxes; Intangibles Tax. Borrower agrees to indemnify and hold Lender and each of its directors, officers, employees, agents, professional advisers and representatives ("Indemnified Parties") harmless from and against any and all claims, damages, losses, liabilities, costs or expenses whatsoever which Lender or any other Indemnified Party may incur (or which may be claimed against any such Indemnified Party by any Person), including attorneys' fees incurred by any Indemnified Party, arising out of or resulting from: (a) the inaccuracy of any representation or warranty of or with respect to Borrower in this Credit Agreement or the other Loan Documents; (b) the failure of Borrower to perform or comply with any covenant or obligation of Borrower under this Credit Agreement or the other Loan Documents; or (c) the exercise by Lender of any right or remedy set forth in this Credit Agreement or the other Loan Documents, provided that Borrower shall have no obligation to indemnify any Indemnified Party against claims, damages, losses, liabilities, costs or expenses to the extent that a court of competent jurisdiction renders a final non-appealable determination that the foregoing are solely the result of the willful

38

misconduct or gross negligence of such Indemnified Party. In addition, Borrower agrees to indemnify and hold the Indemnified Parties harmless from and against any and all claims, damages, losses, liabilities, costs or expenses whatsoever which Lender or any other Indemnified Party may incur (or which may be claimed against any such Indemnified Party by any Person), including attorneys' fees incurred by any Indemnified Party, arising out of or resulting from the imposition or nonpayment by Borrower of any stamp tax, intangibles tax, or similar tax imposed by any state, including any amounts owing by virtue of the assertion that the property valuation used to calculate any such tax was understated. Borrower shall have the right to assume the defense of any claim as would give rise to Borrower's indemnification obligation under this Section with counsel of Borrower's choosing so long as such defense is being diligently and properly conducted and Borrower shall establish to the Indemnified Party's satisfaction that the amount of such claims are not, and will not be, material in comparison to the liquid and unrestricted assets of Borrower available to respond to any award which may be granted on account of such claim. So long as the conditions of the preceding sentence are met, Indemnified Party shall have no further right to reimbursement of attorney's fees incurred thereafter. The obligation to indemnify set forth in this Section shall survive the termination of this Credit Agreement and other covenants.

11.2 <u>Indemnification Relating to Hazardous Substances</u>. Borrower shall not locate, produce, treat, transport, incorporate, discharge, emit, release, deposit or dispose of any Hazardous Substance in, upon, under, over or from any property owned or held by Borrower, except in accordance with all Environmental Laws and Environmental Regulations; Borrower shall not permit any Hazardous Substance to be located, produced, treated, transported, incorporated, discharged, emitted, released, deposited, disposed of or to escape in, upon, under, over or from any property owned or held by Borrower, except in accordance with Environmental Laws and Environmental Regulations; and Borrower shall comply with all Environmental Laws and Environmental Regulations which are applicable to such property. If Lender reasonably believes that an Environmental Law or Environmental Regulation has been violated by Borrower's activities upon property owned or held by Borrower, and if Lender so requests, Borrower shall arrange for the preparation of an environmental review, audit, assessment and/or report relating to the subject property, at Borrower's sole cost and expense, by an engineer or other environmental expert acceptable to Lender. Borrower shall indemnify the Indemnified Parties against, and shall reimburse the Indemnified Parties for, any and all claims, demands, judgments, penalties, liabilities, costs, damages and expenses, including court costs and attorneys' fees incurred by the Indemnified Parties (prior to trial, at trial and on appeal) in any action against or involving the Indemnified Parties, resulting from any breach of the foregoing covenants in this Section or the covenants in Section 9.5 hereof, or from the discovery of any Hazardous Substance in, upon, under or over, or emanating from, such property, it being the intent of Borrower and the Indemnified Parties that the Indemnified Parties shall have no liability or responsibility for damage or injury to human

health, the environmental or natural resources caused by, for abatement and/or clean-up of, or otherwise with respect to, Hazardous Substances by virtue of the interest of Lender in the property created by any documents securing Bank Debt (including without limitation the Loan Documents) or as the result of Lender exercising any of its rights or remedies with respect thereto, including but not limited to becoming the owner thereof by foreclosure or conveyance in lieu of foreclosure; provided that such indemnification as it applies to the exercise by Lender of its rights or remedies with respect to the Loan Documents shall not apply to claims arising solely with respect to

Hazardous Substances brought onto such property by Lender or its successors or assigns while engaged in activities other than operations substantially the same as the operations previously conducted on such property by Borrower. The foregoing covenants of this Section shall be deemed continuing covenants for the benefit of the Indemnified Parties, and any successors and assigns of the Indemnified Parties, including but not limited to the holder of any certificate of purchase, any transferee of the title of Lender or any subsequent owner of the property, and shall survive the satisfaction or release of any lien, any foreclosure of any lien and/or any acquisition of title to the property or any part thereof by Lender, or anyone claiming by, through or under Lender or Borrower by deed in lieu of foreclosure or otherwise. Any amounts covered by the foregoing indemnification shall bear interest from the date incurred at the Default Interest Rate, shall be payable on demand, and shall be secured by the Security Documents. The indemnification and covenants of this Section shall survive the termination of this Credit Agreement and other covenants.

ARTICLE 12. EVENTS OF DEFAULT; RIGHTS AND REMEDIES

- 12.1 Events of Default. The occurrence of any of the following events (each an "Event of Default") shall, at the option of Lender, make the entire Bank Debt immediately due and payable (provided, that in the case of an Event of Default under Subsection 12.1(e) all amounts owing under the Revolving Note and the other Loan Documents shall automatically and immediately become due and payable without any action by or on behalf of Lender), and Lender may exercise all rights and remedies for the collection of any amounts outstanding hereunder and take whatever action it deems necessary to secure itself, all without notice of default, presentment or demand for payment, protest or notice of nonpayment or dishonor, or other notices or demands of any kind or character:
- (a) Failure of Borrower to pay within five (5) days of the due date, whether by acceleration or otherwise, any principal or interest owing under the Revolving Note or any other Bank Debt in accordance with this Credit Agreement or the other Loan Documents.
- (b) Any representation or warranty set forth in any Loan Document, in any Borrowing Notice, in any financial statements or reports or projections or forecasts, or in connection with any transaction contemplated by

40

any such document, shall prove in any material respect to have been false or misleading when made by Borrower.

- (c) Any default by Borrower in the performance or compliance with the covenants, promises, conditions or provisions of Sections 9.7, 9.11, 9.14, 10.1, 10.2, 10.3, 10.4, 10.5, 10.7, 10.8, 10.9, 10.10, or 10.12 of this Credit Agreement.
- (d) The failure of Borrower to pay when due, or failure to perform or observe any other obligation or condition with respect to any of the following obligations to any Person, beyond any period of grace under the instrument creating such obligation: (i) any indebtedness for borrowed money or for the deferred purchase price of property or services, (ii) any obligations under leases which have or should have been characterized as capitalized leases, as determined in accordance with GAAP, or (iii) any contingent liabilities, such as guaranties, for the obligations of others relating to indebtedness for borrowed money or for the deferred purchase price of property or services or relating to obligations under leases which have or should have been characterized as capitalized leases, as determined in accordance with GAAP; provided that no such failure will be deemed to be an Event of Default hereunder unless and until the aggregate amount owing under obligations with respect to which such failures have occurred and are continuing is at least \$200,000.00.
- (e) Borrower, or any of them, applies for or consents to the appointment of a trustee or receiver for any part of its properties; any bankruptcy, reorganization, debt arrangement, dissolution or liquidation proceeding is commenced or consented to by Borrower or any of them; or any application for appointment of a receiver or a trustee, or any proceeding for bankruptcy, reorganization, debt management or liquidation is filed for or commenced against Borrower or any of them, and is not withdrawn or dismissed within sixty (60) days thereafter.
- (f) Failure of Borrower to comply with any other provision of this Credit Agreement or the other Loan Documents not constituting an Event of Default under any of the preceding subparagraphs of this Section 12.1, and such failure continues for thirty (30) days after Borrower learns of such failure to comply, whether by Borrower's own discovery or through notice from Lender.
- (g) The entry of one or more judgments in an aggregate amount in excess of \$200,000.00 against Borrower not stayed, discharged or paid within thirty (30) days after entry.
- 12.2 <u>No Revolving Advance or Letter of Credit.</u> Lender may suspend the Revolving Commitment upon the occurrence and during the continuance of Potential Default or an Event of Default and shall have no obligation to make any Revolving Advance and no obligation to issue or renew any Letter of Credit if a Potential Default or an Event of Default shall occur and be continuing.

41

12.3 <u>Rights and Remedies</u>. In addition to the remedies set forth in Section 12.1 and 12.2 hereof, upon the occurrence of an Event of Default (and unless and until such Event of Default is waived in writing by Lender), Lender shall be entitled to (a) demand that Borrower to establish the Cash Collateral Account and to deposit therein immediately available funds in the amount of the undrawn face amount of all Letters of Credit then outstanding (and if Borrower fails to establish such account and/or provide such funds within two (2) Banking Days of such demand, Lender may establish such account and/or provide such funds by a Revolving Advance, without regard to any limitations thereon); and (b) exercise all the rights and remedies provided in the Security Documents and other Loan

Documents and by any applicable law, including, without limitation, the Uniform Commercial Code as enacted in the state of Colorado or the state which governs perfection of a security interest in the Collateral, whichever provides Lender with greater rights. Each and every right or remedy granted to Lender pursuant to this Credit Agreement and the other Loan Documents, or allowed Lender by law or equity, shall be cumulative. Failure or delay on the part of Lender to exercise any such right or remedy shall not operate as a waiver thereof. Any single or partial exercise by Lender of any such right or remedy shall not preclude any future exercise thereof or the exercise of any other right or remedy.

ARTICLE 13. MISCELLANEOUS

- Costs and Expenses. To the extent permitted by law, Borrower agrees to pay to Lender, on demand, all out-of-pocket costs and expenses (a) incurred by Lender (including, without limitation, the reasonable fees and expenses of counsel retained by Lender, and including fees and expenses incurred for consulting, appraisal, engineering, inspection, and environmental assessment services) in connection with the preparation, negotiation, and execution of the Loan Documents and the transactions contemplated thereby, and processing each Borrowing Notice; and (b) incurred by Lender (including, without limitation, the reasonable fees and expenses of counsel retained by Lender) in connection with the enforcement or protection of Lender's rights under the Loan Documents upon the occurrence of an Event of Default or upon the commencement of an action by Borrower against Lender (except that if the court makes a specific finding that Borrower has prevailed on all or substantially all of its claims in such action brought by Borrower, Borrower shall not be obligated to pay the costs and expenses of Lender in connection with such action), including without limitation collection of the Revolving Loan and enforcement of its rights with respect to the Collateral (regardless of whether such enforcement or collection is by court action or otherwise). Lender may, at its discretion, pay all amounts owing pursuant to clause (a) above by a Revolving Advance upon notice to Gaiam.
- 13.2 <u>Service of Process and Consent to Jurisdiction</u>. Borrower hereby agrees that any litigation with respect to this Credit Agreement or to enforce any judgment obtained against Borrower for breach of this Credit Agreement or

42

under the Revolving Note or other Loan Documents may be brought in the courts of the State of Colorado and in the United States District Court for the District of Colorado (if applicable subject matter jurisdictional requirements are present), as Lender may elect; and, by execution and delivery of this Credit Agreement, Borrower irrevocably submits to such jurisdiction. With respect to litigation concerning this Credit Agreement or under the Revolving Note or other Loan Documents within the jurisdiction of the courts of the State of Colorado or the United States District Court for the District of Colorado, Borrower hereby irrevocably appoints, until at least one (1) year following the Revolving Maturity Date, Corporation Service Company to serve as agent of Borrower to receive for and on behalf of Borrower at such agent's Denver, Colorado office (presently at 1560 Broadway), service of process, which service may be made by mailing a copy of any summons or other legal process to Borrower in care of such agent. Borrower agrees that Borrower shall maintain a duly appointed agent in Colorado for service of summons and other legal process as long as Borrower remains obligated under this Credit Agreement and shall keep Lender advised in writing of the identity and location of such agent. The receipt by such agent and/or by Borrower of such summons or other legal process in any such litigation shall be deemed personal service and acceptance by Borrower for all purposes of such litigation.

- 13.3 <u>Jury Waiver</u>. IT IS MUTUALLY AGREED BY AND BETWEEN LENDER AND BORROWER THAT THEY EACH WAIVE TRIAL BY JURY IN ANY ACTION, PROCEEDING, OR COUNTERCLAIM BROUGHT BY EITHER OF THEM AGAINST THE OTHER PARTY ON ANY MATTER WHATSOEVER ARISING OUT OF OR IN ANY WAY CONNECTED WITH THIS CREDIT AGREEMENT, THE REVOLVING NOTE, OR THE OTHER LOAN DOCUMENTS.
- Arbitration. Notwithstanding the provisions of Sections 13.2 and 13.3 hereof, the parties hereto agree, upon demand by either party, to submit to binding arbitration all claims, disputes and controversies between or among them (and their respective employees, officers, directors, attorneys, and other agents), whether in tort, contract or otherwise arising out of or relating to in any way (a) the Loans and related Loan Documents which are the subject of this Credit Agreement and its negotiation, execution, collateralization, administration, repayment, modification, extension, substitution, formation, inducement, enforcement, default or termination; or (b) requests for additional credit.
- 13.4.1 Governing Rules. Any arbitration proceeding will (a) proceed in a location in Colorado selected by the American Arbitration Association ("AAA"); (b) be governed by the Federal Arbitration Act (Title 9 of the United States Code), notwithstanding any conflicting choice of law provision in any of the documents between the parties; and (c) be conducted by the AAA, or such other administrator as the parties shall mutually agree upon, in accordance with the AAA's commercial dispute resolution procedures, unless the claim or counterclaim is at least \$1,000,000.00 exclusive of claimed interest,

43

arbitration fees and costs in which case the arbitration shall be conducted in accordance with the AAA's optional procedures for large, complex commercial disputes (the commercial dispute resolution procedures or the optional procedures for large, complex commercial disputes to be referred to, as applicable, as the "Rules"). If there is any inconsistency between the terms hereof and the Rules, the terms and procedures set forth herein shall control. Any party who fails or refuses to submit to arbitration following a demand by any other party shall bear all costs and expenses incurred by such other party in compelling arbitration of any dispute. Nothing contained herein shall be deemed to be a waiver by any party that is a bank of the protections afforded to it under 12 U.S.C. §91 or any similar applicable state law.

13.4.2 No Waiver of Provisional Remedies, Self-Help and Foreclosure. The arbitration requirement does not limit the right of any party to (a) foreclose against real or personal property collateral; (b) exercise self-help remedies relating to collateral or proceeds of collateral such as setoff or repossession; or (c) obtain provisional or ancillary remedies such as replevin, injunctive relief, attachment or the appointment of a receiver, before during or after the pendency of any arbitration proceeding. This exclusion does not constitute a waiver of the right or obligation of any party to submit any dispute to arbitration or reference hereunder, including those arising from the exercise of the actions detailed in sections (a), (b) and (c) of this Subsection.

13.4.3 <u>Arbitrator Qualifications and Powers</u>. Any arbitration proceeding in which the amount in controversy is \$5,000,000.00 or less will be decided by a single arbitrator selected according to the Rules, and who shall not render an award of greater than \$5,000,000.00. Any dispute in which the amount in controversy exceeds \$5,000,000.00 shall be decided by majority vote of a panel of three arbitrators; provided however, that all three arbitrators must actively participate in all hearings and deliberations. The arbitrator will be a neutral attorney licensed in the State of Colorado or a neutral retired judge of the state or federal judiciary of Colorado, in either case with a minimum of ten years experience in the substantive law applicable to the subject matter of the dispute to be arbitrated. The arbitrator will determine whether or not an issue is arbitratable and will give effect to the statutes of limitation in determining any claim. In any arbitration proceeding the arbitrator will decide (by documents only or with a hearing at the arbitrator's discretion) any pre-hearing motions which are similar to motions to dismiss for failure to state a claim or motions for summary adjudication. The arbitrator shall resolve all disputes in accordance with the substantive law of Colorado and may grant any remedy or relief that a court of such state could order or grant within the scope hereof and such ancillary relief as is necessary to make effective any award. The arbitrator shall also have the power to award recovery of all costs and fees, to impose sanctions and to take such other action as the arbitrator deems necessary to the same extent a judge could pursuant to the Federal Rules of Civil Procedure, the Colorado Rules of Civil Procedure or other applicable law. Judgment upon the award rendered by the arbitrator may be entered in any court having jurisdiction. The

44

institution and maintenance of an action for judicial relief or pursuit of a provisional or ancillary remedy shall not constitute a waiver of the right of any party, including the plaintiff, to submit the controversy or claim to arbitration if any other party contests such action for judicial relief.

13.4.4 <u>Discovery</u>. In any arbitration proceeding discovery will be permitted in accordance with the Rules. All discovery shall be expressly limited to matters directly relevant to the dispute being arbitrated and must be completed no later than twenty (20) days before the hearing date and within 180 days of the filing of the dispute with the AAA. Any requests for an extension of the discovery periods, or any discovery disputes, will be subject to final determination by the arbitrator upon a showing that the request for discovery is essential for the party's presentation and that no alternative means for obtaining information is available.

13.4.5 <u>Class Proceedings and Consolidations</u>. The resolution of any dispute arising pursuant to the terms of this Credit Agreement shall be determined by a separate arbitration proceeding and such dispute shall not be consolidated with other disputes or included in any class proceeding.

13.4.6 <u>Payment of Arbitration Costs and Fees</u>. The arbitrator shall award all costs and expenses of the arbitration proceeding.

13.4.7 <u>Miscellaneous</u>. To the maximum extent practicable, the AAA, the arbitrators and the parties shall take all action required to conclude any arbitration proceeding within 180 days of the filing of the dispute with the AAA. No arbitrator or other party to an arbitration proceeding may disclose the existence, content or results thereof, except for disclosures of information by a party required in the ordinary course of its business or by applicable law or regulation. If more than one agreement for arbitration by or between the parties potentially applies to a dispute, the arbitration provision most directly related to the Loan Documents or the subject matter of the dispute shall control. This arbitration provision shall survive termination, amendment or expiration of any of the Loan Documents or any relationship between the parties.

Notices. All notices, requests and demands required or permitted under the terms of this Credit Agreement shall be in writing and (a) shall be addressed as set forth below or at such other address as either party shall designate in writing, (b) shall be deemed to have been given or made: (i) if delivered personally, immediately upon delivery, (ii) if by facsimile transmission, immediately upon sending and upon confirmation of receipt, (iii) if by nationally recognized overnight courier service with instructions to deliver the next Banking Day, one (1) Banking Day after sending, and (iv) if by United States Mail, certified mail, return receipt requested, five (5) Banking Days after mailing.

45

13.5.1 <u>Borrower</u>:

Gaiam, Inc. Suite 300 360 Interlocken Boulevard Broomfield, Colorado 80021

FAX: (303) 265-9110

Attention: Chief Financial Officer

13.5.2 Lender:

Wells Fargo Bank, National Association 1242 Pearl Street P.O. Box 227 Boulder, Colorado 80302

FAX: (303) 444-3295

Attention: Gaiam Relationship Manager

- 13.6 Successors and Assigns. This Credit Agreement shall be binding upon and inure to the benefit of Borrower and Lender and their respective successors and assigns, except that Borrower may not assign or transfer its rights or obligations hereunder without the prior written consent of Lender. Lender may, without Borrower's consent, (a) transfer its interest hereunder to a Person at least fifty percent (50%) the equity interest in which is owned by Lender or which owns at least fifty percent (50%) of the equity interest of which is owned by the same Person which owns at least fifty percent (50%) of the equity interest of Lender; and (b) grant to a Person one or more participation interests in the Loans. In connection with any such assignment or transfer, or negotiations in connection with any proposed assignment, transfer, or participation, Lender may disclose all documents and information which Lender now has or hereafter acquires relating to Borrower or its business or any Collateral.
- 13.7 <u>Severability.</u> The invalidity or unenforceability of any provision of this Credit Agreement or the other Loan Documents shall not affect the remaining portions of such documents or instruments; in case of such invalidity or unenforceability, such documents or instruments shall be construed as if such invalid or unenforceable provisions had not been included therein.
- 13.8 <u>Applicable Law.</u> To the extent not governed by federal law, this Credit Agreement and the other Loan Documents, and the rights and obligations of the parties hereto and thereto shall be governed by and interpreted in accordance with the internal laws of the State of Colorado, without giving effect to any otherwise applicable rules concerning conflicts of law.

46

- 13.9 <u>Captions</u>. The captions or headings in this Credit Agreement and any table of contents hereof are for convenience only and in no way define, limit or describe the scope or intent of any provision of this Credit Agreement.
- 13.10 Complete Agreement; Amendments. THIS CREDIT AGREEMENT, THE REVOLVING NOTE, AND THE OTHER LOAN DOCUMENTS ARE INTENDED BY THE PARTIES HERETO TO BE A COMPLETE AND FINAL EXPRESSION OF THEIR AGREEMENT AND MAY NOT BE CONTRADICTED BY EVIDENCE OF ANY PRIOR OR CONTEMPORANEOUS ORAL AGREEMENT. LENDER AND BORROWER ACKNOWLEDGE AND AGREE THAT NO UNWRITTEN ORAL AGREEMENT EXISTS BETWEEN THEM WITH RESPECT TO THE SUBJECT MATTER OF THIS AGREEMENT. This Credit Agreement (together with all exhibits hereto, which are incorporated herein by this reference) and the other Loan Documents represent the entire understanding of Lender and Borrower with respect to the subject matter hereof and shall replace and supersede any previous agreements of the parties with respect to the subject matter hereof. This Credit Agreement may not be modified or amended unless such modification or amendment is in writing and is signed by Borrower and Lender. Borrower agrees that it shall reimburse Lender for all reasonable fees and expenses incurred by Lender in retaining outside legal counsel in connection with any amendment or modification to this Credit Agreement requested by Borrower.
- 13.11 Additional Costs of Maintaining Loan. Borrower shall pay to Lender from time to time such amounts as Lender may determine to be necessary to compensate Lender for any increase in costs to Lender which Lender determines are attributable to Lender's making or maintaining a Revolving Advance hereunder or its obligation to make such Revolving Advance, or any reduction in any amount receivable by Lender under this Credit Agreement or the Revolving Note in respect to such Revolving Advance or such obligation (such increases in costs and reductions in amounts receivable being herein called "Additional Costs"), resulting from any change after the date of this Credit Agreement in United States federal, state, municipal laws or regulations (including Regulation D of the Federal Reserve Board), or the adoption or making after such date of any interpretations, directives, or requirements applying to a class of banks including Lender of or under any United States federal, state, municipal laws or regulations (whether or not having the force of law) by any court or governmental or monetary authority charged with the interpretation or administration thereof ("Regulatory Change"), which: (a) changes the basis of taxation of any amounts payable to Lender under this Credit Agreement or the Revolving Note in respect of such Revolving Advance (other than taxes imposed on the overall net income of Lender); or (b) imposes or modifies any reserve, special deposit, or similar requirements relating to any extensions of credit or other assets of, or any deposits with or other liabilities of, Lender; or (c) imposes any other condition affecting this Credit Agreement or the Revolving Note (or any of such extensions of credit or liabilities). Lender will notify Borrower of any event occurring after the date of this Credit Agreement

47

which will entitle Lender to compensation pursuant to this Section as promptly as practicable after it obtains knowledge thereof and determines to request such compensation. Lender shall include with such notice, a certificate from Lender setting forth in reasonable detail the calculation of the amount of such compensation. Determinations by Lender for purposes of this Section of the effect of any Regulatory Change on the costs of Lender of making or maintaining a Revolving Advance or on amounts receivable by Lender in respect of Revolving Advances, and of the additional amounts required to compensate Lender in respect of any Additional Costs, shall be conclusive absent manifest error, provided that such determinations are made on a reasonable basis.

- 13.12 <u>Capital Requirements.</u> In the event that the introduction of or any change in: (a) any law or regulation; or (b) the judicial, administrative, or other governmental interpretation of any law or regulation; or (c) compliance by Lender or any corporation controlling Lender with any guideline or request from any governmental authority (whether or not having the force of law) has the effect of requiring an increase in the amount of capital required or expected to be maintained by Lender or any corporation controlling Lender, and Lender certifies that such increase is based in any part upon Lender's obligations hereunder, and other similar obligations, Borrower shall pay to Lender such additional amount as shall be certified by Lender to Borrower to be the net present value (discounted at the Base Rate less the Base Rate Margin) of the amount by which such increase in capital reduces the rate of return on capital which Lender could have achieved over the period remaining until the Revolving Maturity Date but for such introduction or change. Lender will notify Borrower of any event occurring after the date of this Credit Agreement that will entitle Lender to compensation pursuant to this Section as promptly as practicable after it obtains knowledge thereof and of Lender's determination to request such compensation. Lender shall include with such notice, a certificate setting forth in reasonable detail the calculation of the amount of such compensation. Determinations by Lender for purposes of this Section of the effect of any increase in the amount of capital required to be maintained by Lender and of the amount of compensation owed to any Lender under this Section shall be conclusive absent manifest error, provided that such determinations are made on a reasonable basis.
- 13.13 <u>Replacement Notes.</u> Upon receipt by Borrower of evidence reasonably satisfactory to it of the loss, theft, destruction or mutilation of any Revolving Note, and (in case of loss, theft or destruction) of the agreement of Lender to indemnify Borrower, and upon surrender and cancellation of such Revolving Note, if mutilated, then Borrower will execute and deliver a new Revolving Note to be dated as of the date of this Credit Agreement.
- 13.14 <u>Mutual Release</u>. Upon full indefeasible payment and satisfaction of the Bank Debt and Revolving Note and the other obligations contained in this Credit Agreement, Borrower and Lender shall, except as provided in Article 11 hereof with respect to indemnification obligations, thereupon automatically each

48

be fully, finally, and forever released and discharged from any further claim, liability, or obligation in connection with the Bank Debt.

- 13.15 No Implied Waiver. The acceptance by Lender at any time and from time to time of partial payments on the Bank Debt shall not be deemed a waiver of any Potential Default or Event of Default then existing. No waiver by Lender of any Potential Default or Event of Default shall be deemed to be a waiver of any other then-existing or subsequent Potential Default or event of Default. No delay or omission by Lender in exercising any right or remedy shall impair that right or remedy or be construed as a waiver thereof, nor shall any single or partial exercise of any right or remedy preclude other or further exercise of that, or any other, right or remedy.
- 13.16 <u>Liberal Construction</u>. This Credit Agreement constitutes a fully negotiated agreement between commercially sophisticated parties, each assisted by legal counsel, and shall not be construed and interpreted for or against any party hereto.
- 13.17 Counterparts. This Credit Agreement may be executed in any number of counterparts and by different parties to this Credit Agreement in separate counterparts, each of which when so executed shall be deemed to be an original and all of which taken together shall constitute one and the same agreement. Copies of documents or signature pages bearing original signatures, and executed documents or signature pages delivered by facsimile or by e-mail transmission of an Adobe® file format document (also known as a PDF file), shall, in each such instance, be deemed to be, and shall constitute and be treated as, an original signed document or counterpart, as applicable. Any party delivering an executed counterpart of this Credit Agreement by facsimile, or by e-mail transmission of an Adobe file format document also shall deliver an original executed counterpart of this Credit Agreement, but the failure to deliver an original executed counterpart shall not affect the validity, enforceability, and binding effect of this Credit Agreement.
- 13.18 Affect of Amended and Restated Credit Agreement. This Credit Agreement shall, upon compliance with the requirements of Section 8.1 hereof, be effective from the Effective Date forward, and the execution of this Credit Agreement shall not relieve any party to the 2002 Credit Agreement from their respective obligations thereunder for the period from the Original Effective Date to the Effective Date or from any liability for the failure to perform such obligations or from any liability arising out of indemnification obligations under the 2002 Credit Agreement.

[SIGNATURES BEGIN ON THE NEXT PAGE]

49

IN WITNESS WHEREOF, the parties have executed this Credit Agreement as of the date first above written.

LENDER:

Wells Fargo Bank, National Association

By: /s/ Zan M. Powell
Name: Zan M. Powell
Title: Vice President
Applicable Lending Office:
1242 Pearl Street
Boulder, CO 80302

BORROWER:

Gaiam, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Lynn Powers
Name: Lynn Powers
Title: President

Gaiam Americas, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Lynn Powers
Name: Lynn Powers
Title: President

Gaiam.com, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Lynn Powers Name: Lynn Powers Title: President

Gaiam Direct, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Lynn Powers Name: Lynn Powers Title: President

50

Gaiam International, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Janet Mathews
Name: Janet Mathews
Title: Secretary/Treasurer

Gaiam International II, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Janet Mathews
Name: Janet Mathews
Title: Secretary/Treasurer

Gaiam International III, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Janet Mathews
Name: Janet Mathews
Title: Secretary/Treasurer

Gaiam Media, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Lynn Powers
Name: Lynn Powers
Title: President

Gaiam Shared Services, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Lynn Powers

Gaiam Travel, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Lynn Powers
Name: Lynn Powers
Title: President

Gaiam West, Inc., a corporation formed under the laws of the State of California.

By: /s/ Lynn Powers Name: Lynn Powers Title: President

GT Media, Inc., a corporation formed under the laws of the State of Colorado.

By: /s/ Lynn Powers
Name: Lynn Powers
Title: President

CERTIFICATION

I, Jirka Rysavy, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Gaiam, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15((f)) for the registrant and have:
 - (a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with the generally accepted accounting principles;
 - (c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) all significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 9, 2005

/s/ Jirka Rysavy
Jirka Rysavy
Chairman of the Board and

Chief Executive Officer

CERTIFICATION

I, Janet Mathews, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Gaiam, Inc.;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) for the registrant and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15((f)) for the registrant and have:
 - (a) designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with the generally accepted accounting principles;
 - (c) evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

Date: August 9, 2005 /s/ Janet Mathews

Janet Mathews Chief Financial Officer

CERTIFICATIONS PURSUANT TO 18 U.S.C. SECTION 1350

In connection with the Quarterly Report of Gaiam, Inc. (the "Company") on Form 10-Q for the period ending June 30, 2005, as filed with the U.S. Securities and Exchange Commission on the date hereof (the "Report"), I, Jirka Rysavy, Chairman and Chief Executive Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that to my knowledge:

- (1) The Report fully complies with the requirements of Sections 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 9, 2005

/s/ Jirka Rysavy

Jirka Rysavy Chairman of the Board and Chief Executive Officer

CERTIFICATIONS PURSUANT TO 18 U.S.C. SECTION 1350

In connection with the Quarterly Report of Gaiam, Inc. (the "Company") on Form 10-Q for the period ending June 30, 2005, as filed with the U.S. Securities and Exchange Commission on the date hereof (the "Report"), I, Janet Mathews, Chief Financial Officer of the Company, certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that to my knowledge:

- (1) The Report fully complies with the requirements of Sections 13(a) or 15(d) of the Securities Exchange Act of 1934; and
- (2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

Date: August 9, 2005

/s/ Janet Mathews

Janet Mathews Chief Financial Officer